Contents lists available at ScienceDirect

Energy Economics

journal homepage: www.elsevier.com/locate/eneeco

Dynamics between global value chain participation, CO₂ emissions, and economic growth: Evidence from a panel vector autoregression model

Jing Wang^a, Dan S. Rickman^b, Yihua Yu^{c,*}

^a School of Economics, Chongqing Technology and Business University, Chongqing 400067, China

^b Department of Economics, Oklahoma State University, Stillwater, OK 74078, USA

^c School of Applied Economics, Renmin University of China, Beijing 100872, China

ARTICLE INFO

JEL codes: G14 F23 O40 O44 Q56 Keywords: global value chain participation CO₂ emission economic growth panel vector autoregressive model sustainable growth global warming

ABSTRACT

Resolving the conflict between economic growth and CO_2 reduction is critical for sustainable growth. Increasing integration into global value chains (GVCs) is an inevitable trend for countries to develop their international markets. However, the dynamic relationships between GVC participation, CO_2 emissions, and economic growth have not been fully clarified. This study analyzes these relationships and the underlying mechanisms using a panel vector autoregressive model approach with data for 63 countries and regions from 2005 to 2015. The major findings are: (1) GVC participation promotes environmentally-friendly growth in the long run by increasing per capita gross domestic product (GDP) and reducing per capita CO_2 emissions. (2) GVC participation increases as per capita CO_2 emissions increase; it decreases as per capita GDP grows. (3) These relationships vary by industry and income. The variation of GVC participation in high- CO_2 emission industries explains a large proportion of the variation in per capita GDP. Further, high-income countries benefit more from GVC participation compared to low-income countries. An important policy recommendation is that countries should actively participate in GVCs to promote sustainable growth.

1. Introduction

The disappearance of Okjökull, the first Icelandic glacier lost to climate change, in 2014, once again reminded the world that global warming cannot be ignored. "In the next 200 years all our glaciers are expected to follow the same path," reads the plaque commemorating the loss of Okjökull. Greenhouse gas is the well-known culprit of global warming. CO2 emissions account for about two-thirds of global greenhouse gas emissions (Oliver et al., 2016). In recent decades, global CO2 emissions have continued to increase. From 1990 to 2018, the overall CO₂ emissions and CO₂ per capita grew by 61% and 12%, respectively (Fig. 1). Consequently, CO₂ emissions have become a crucial worldwide topic in studies on climate change and environmentally-friendly growth (Tol, 2005; Raupach et al., 2007; Chen et al., 2019). During this period, the world economy and international trade have also developed significantly. Clarifying the relationships between CO₂ emissions, economic growth, and international trade is of great importance to sustainable growth. The emergence of global value chains (GVCs), resulting from

vertical specialization and worldwide intra-industry trade, complicates these relationships.

Per capita income is the primary driver of CO₂ emissions (Parker and Bhatti, 2020), largely because of energy consumption during the production process (Wang et al., 2005; Su and Ang, 2012; Wang et al., 2014; Xu et al., 2014; Zhang and Da, 2015). Energy consumption varies significantly across industries, which implies that industry composition is one of the determinants of CO₂ emissions. Fig. 2 shows that the electricity and heat production sector produced the largest share of global CO2 emissions during the period 1990 to 2014, followed by transport, manufacturing and construction, residential buildings and commercial and public services, and other sectors. Therefore, besides improving energy efficiency, a way for a country to reduce its CO₂ emissions would be to curb domestic production in CO2-generating industries, transfer them overseas, and import the products of these environmentally unfriendly industries to meet the domestic demand. This process is popularly known as emission transfer via international trade (Kleemann and Abdulai, 2013; Aklin, 2016).¹ In other words,

* Corresponding author.

https://doi.org/10.1016/j.eneco.2022.105965

Received 31 October 2020; Received in revised form 28 February 2022; Accepted 7 March 2022 Available online 18 March 2022 0140-9883/© 2022 Elsevier B.V. All rights reserved.





E-mail addresses: jing.wang@ctbu.edu.cn (J. Wang), dan.rickman@okstate.edu (D.S. Rickman), yihua.yu@ruc.edu.cn (Y. Yu).

 $^{^{1}}$ It is worth noting that emission transfer may not reduce global CO₂ emissions if it simply transfers emissions from one country to another.



Fig. 1. Carbon emissions (million tons, left axis) and per capita carbon emissions (tons/person, right axis) from fossil fuel combustion and cement production, 1990-2018

Data Sources: Carbon emissions data come from The Global Carbon Budget 2019; population size data used to compute per capita carbon emissions come from the World Bank Database

Note: 1 million tons carbon = 3.664 million tons of CO₂.



II Manufacturing industries and construction

Fig. 2. Industry composition of global CO₂ emissions, 1990–2014 (%). Data Source: The World Bank Database.

international trade plays a critical role in CO2 emissions and CO2 redistributions across countries and regions.

As intra-industry trade plays an increasingly key role in international trade, the world economy has entered the GVC era (Wang et al., 2019). Within the framework of GVCs, the production processes are distributed globally (Zhang et al., 2017), and countries are responsible for one or more sections of GVCs instead of completing the whole production process independently. The emergence of GVCs makes the relationships between trade, CO₂ emissions, and economic growth more complicated. Fig. 3 shows the GVC participation degrees and per capita CO₂ emissions of 63 countries and regions from 2005 to 2015.² Based on the World Bank Analytical Classifications 2005–2015, we divided these countries into four groups: high-income (H), upper-middle-income (UM), lowermiddle-income (LM), and low-income (L). An observation of Fig 3(a) and 3(b) revealed that: (1) The average GVC participation degree of high-income countries was generally higher than others, as were average per capita CO2 emissions. Moreover, per capita CO2 emissions increased as income levels increased, while there was neither a

Electricity and heat production

continuous positive nor negative correlation between GVC participation and income. (2) After experiencing a slump during the 2008-2009 financial crisis and a subsequent recovery period in 2010-2011, except for the low-income group, GVC participation showed a downward trend after 2011. (3) The per capita CO_2 emissions of the high-income group decreased over time, while those of other groups showed a slight upward trend after 2010.

Four effects-scale effect, technique spillover effect, composition effect, and competition effect-simultaneously act on the GVC-CO2 nexus (Wang et al., 2019). Thus, GVC participation has differentiated impacts on CO₂ emissions across countries and sectors. Similarly, the impact of integration into GVCs on economic growth differs across countries and sectors (Jouanjean et al., 2017). Additionally, the reverse effects of CO₂ emissions and economic growth on GVC participation cannot be ignored. On the one hand, high CO2 emissions imply relatively weak environmental stringency, thus promoting the production of environmentally unfriendly products in a country and its GVC participation in relative industries. On the other hand, high CO₂ emissions hurt the ecosystem, forcing a country to take measures to reduce emissions, such as shutting down polluting factories, which will thwart its production in related industries and the level of GVC participation.

These facts lead to the following questions: (1) How do GVC participation, CO2 emissions, and economic growth affect each other? (2) What are the influencing mechanisms? (3) How can a country

 $^{^{2}}$ On average, these countries accounted for 84.54%, 92.09%, 89.27%, and 88.58% of the global CO₂ emissions, GDP, imports, and exports, respectively, for the period of study. GDP, import value, and export value were measured in constant 2010 USD.



(a) GVC participation degree

(b) Per capita CO₂ emissions

Fig. 3. GVC participation degrees (%) and per capita CO₂ emissions (tons/person) by country income group, 2005–2015. *Data Sources:* The carbon emissions data come from Trade in Value Added Database (2018 edition); the population size data used to compute per capita carbon emissions come from the World Bank Database.

Notes: 1. H, UM, LM, and L stand for high-income, upper-middle-income, lower-middle-income, and low-income groups. 2. According to the World Bank Analytical Classification 2015, no country in our sample belonged to the low-income group in 2015.

achieve environmentally-friendly growth in the context of the international vertical division and cooperation? It is expected that people prefer environmentally-friendly growth, not economic growth at the expense of the environment, or a system that is environmentally friendly with higher poverty. However, at present, cleaner production cannot be the spontaneous or preferred behavior of most producers around the world, especially in developing and undeveloped countries and regions. This is largely due to the inaccessibility and high cost of clean energy and clean production technologies. Therefore, the answers to these questions are crucial for policymakers to design effective green development policies and strategies.

To explore these questions, we applied the panel vector autoregression (PVAR) model, using data for 63 countries and regions from 2005 to 2015, and empirically analyzed the dynamic relationships between GVC participation, CO_2 emissions, and economic growth, mainly focusing on the impact of GVC participation on sustainable growth (i.e., environmentally-friendly growth). The PVAR model is widely used to analyze the relationships between endogenous variables using panel data (Love and Zicchino, 2006; Jawadi et al., 2016; Magazzino, 2016a; Magazzino, 2016b; Antonakakis et al., 2017; Magazzino, 2017), fitting with our research.

Although we were unable to study more countries and regions over a longer period due to data limitations and our research focus does not cover spatial correlation or GVC position,³ this study makes three primary contributions. First, to our best knowledge, it is the first study examining the aforementioned dynamic relationships. Previous research focused separately on the impact of GVC participation on CO₂ emissions, the impact of GVC participation on gross domestic product (GDP), or the impact of GDP on CO₂ emissions. The studies failed to explore fully the dynamic relationships and influencing mechanisms of the three variables and the effects of GVC participation on sustainable growth. Our empirical results not only confirm the impacts of GVC participation on CO2 emissions and GDP, and the effect of GDP on CO2 emissions, but also unveil the reverse effects of CO2 emissions and GDP on GVC participation, and the reverse effect of CO2 emissions on GDP. Second, unlike previous studies, we considered the heterogeneous effects of GVC participation in diverse industries. Third, developing and developed countries have been playing divergent roles in GVCs and CO2 emissions, so we examined the heterogeneity in the relationships between highand low-income countries. Our empirical results show that the dynamic relationships vary across industry and income groups. We further explored the influencing mechanisms through heterogeneity analyses.

The remaining sections of this article are organized as follows. Section 2 reviews the relevant literature and develops testable hypotheses. Section 3 introduces the PVAR model and describes the variables and data. Section 4 presents the baseline results and several robustness checks. Section 5 provides further discussion on the dynamics between GVC participation, CO_2 emissions, and economic growth. Section 6 summarizes the main conclusions of the study and discusses policy implications. Limitations and future research are proposed in Section 7.

2. Literature review and hypotheses development

2.1. GVC-CO₂ nexus

Research on the GVC-CO₂ relationship, some of which supports the promoting effect of GVC participation on CO₂ emissions, is scarce. López et al. (2013) studied Spain-China trade in 2005 and found that the pollution haven hypothesis explained 29,667 Kt of CO₂ emissions, 24.1% of which was caused by GVC participation. Hertwich (2020) found that increases in GVC participation drove the rise of carbon-intransit. Meng et al. (2018) suggested a correlation between GVC participation and CO₂ emissions without empirical analysis. Yao et al. (2021) posited that countries with high GVC are more energy-efficient and able to manufacture products with fewer energy inputs, and hence cause fewer emissions. Wang et al. (2019) empirically estimated the impact of GVC participation on per capita CO2 emissions, using panel data of 62 countries from 1995 to 2011. They found an inverted-U relationship between GVC and CO2 and attributed it to the combined results of scale, technique spillover, composition, and competition effects. Lv et al. (2019) also established a nonlinear effect of GVC participation on the carbon emissions embodied in export trade.

Despite the lack of research focused on the impact of CO_2 emissions on GVC participation, we argue that CO_2 emissions can affect GVC participation through the following two channels: (1) *Carbon transfer effect*. Ben-David et al. (2018) pointed out that multinationals that have headquarters in countries with strict environmental policies launch their activities in countries with poor environmental laws. Zhang et al. (2017) showed that low production costs and lower environmental regulations in developing countries lead developed countries to outsource their polluting production processes. The effect of carbon transfer leads to

³ Please refer to limitations and future research of Section 7 for details.

international production specialization. Horizontal specialization enhances inter-industry trade across countries, while vertical specialization stimulates GVC participation. Nearly 70% of the increases in international trade can be attributed to GVC specialization (Yi, 2003). Developed countries with high CO₂ emissions will participate in GVCs to transfer emissions out. Duan et al. (2021) posited that high-income countries transfer their emissions offshore to low-income countries by outsourcing only dirty production stages instead of entire production processes. Some developing countries have enacted relatively loose environmental regulations in search of economic growth (Arce González et al., 2012; Yasmeen et al., 2019). Relatively low CO2 standards and hence relatively high CO2 emissions will bring opportunities for developing countries to participate in GVCs to make use of their comparative advantages in international markets. (2) Environmental protection effect. Companies worldwide have started redefining their business models to overcome environmental challenges and reach sustainable goals (WTO, 2018). A growing number of citizens focus on coordinated development between the environment and economic growth and play increasingly important roles in making decisions on environmental treatments (Do Paco et al., 2009; Glucker et al., 2013). The environmental protection effect pushes countries to participate in clean GVC activities and retreat from dirty GVC activities. Drake-Brockman (2018) emphasized the importance of sustainable GVC-linked investments where low-carbon opportunities are identified.

This discussion leads us to the following hypothesis:

H1. : CO₂ emissions respond to GVC participation and vice versa, and this varies by country.

2.2. GVC-GDP nexus

There are few studies on this relationship and they reported mixed findings. Ignatenko et al. (2019) analyzed 189 countries from 1990 to 2013 and found a positive impact of participation in GVCs on income per capita and productivity. Kowalski et al. (2015) reported that participation in GVCs raised the value added and productivity. However, Rodrik (2018) criticized the contribution of participation in value chains on output and productivity growth. Fagerberg et al. (2018) established that GVC participation failed to raise output growth by observing 125 countries during 1997–2013.

A few studies focused on the differentiated effects of GVC participation on output across countries and sectors. For example, Kummritz (2015) demonstrated the positive impact of GVC participation on domestic value added and that this impact was significant only for middleand high-income countries. The results of Fagerberg et al. (2018) imply that small countries and countries with low capabilities are at a disadvantage in terms of benefitting from value chains. Formai and Vergara Caffarelli (2015) found that participation in GVCs presented differentiated impacts on total factor productivity and labor productivity growth across sectors by analyzing 50 countries from 1990 to 2009. Kordalska et al. (2016) provided estimations for 20 industries (13 manufacturing and seven service industries) in 40 countries. They estimated significantly positive results of backward GVC participation, mainly for the manufacturing industries.

Existing literature rarely paid attention to the reverse effect of GDP on GVC participation, but we argue that GDP can affect GVC participation in return because of three aspects: (1) Productivity effect. Countries that move up in GVCs gain larger benefits from international specialization and global production fragmentation (Liu et al., 2018). High-income countries usually participate in high-tech and capital-intensive industries' GVCs and stay at the upper ends of GVCs due to their comparative advantages at technological innovation and productivity. By outsourcing the production of comparative disadvantages via GVCs, a country can focus on the production in which it has comparative advantages and thus further improve productivity. In addition, high-income countries' comparative advantages are relatively non-

substitutable in international markets. By participating in GVCs, a country can benefit from exports of advanced technologies to the rest of the world. Thus, as income rises, countries are willing and able to participate in the upper ends of GVCs. (2) *Technique spillover effect*. Low-income countries are eager to participate in GVCs to benefit from technique spillover effects (Borck and Coglianese, 2009; Zhou et al., 2020) from high-come countries. (3) *Substitution effect*. Since the comparative advantages of low-income countries usually lie in low costs and/or weak environmental supervision (Krueger, 1977; Chichilnisky, 1994; Erdogan, 2014; Ee et al., 2018), they could generally participate in the labor- and natural-resource intensive industries' GVCs and stay at the lower ends of GVCs. Moreover, their comparative advantages can be substituted relatively easily.⁴ Therefore, as income drops, countries' willingness to participate in GVCs increases but their participation may decline.

Thus, we put forward the following hypothesis:

H2. : GVC participation affects GDP and vice versa, and the effect may vary by country and industry.

2.3. CO₂-GDP nexus

The relationship between CO2 emissions and GDP has long been a concern of scholars. Early studies (see Holtz-Eakin and Selden, 1995; World Bank Development Report, 1992; Selden and Song, 1994; Grossman and Krueger, 1991, 1995) found that the environmental degradation-income relationship presented an inverted U-shape, which is well known as the environmental Kuznets curve (EKC) proposed by Grossman and Krueger (1991). Pollution increases up to a certain level as income goes up; after that, it decreases. This relationship has been tested by many recent studies. For instance, Charfeddine (2017) averred that the EKC hypothesis holds for CO_2 emissions. Ang (2007), using data of France for 1960-2000, found that CO₂ emissions and output had a quadratic relationship in the long run. The empirical results of Jaunky (2011) provided evidence of EKC for Greece, Malta, Oman, Portugal, and the U.K. Zhang and Zhang (2018) verified the validity of the EKC hypothesis for China. You and Lv (2018) found strong evidence for the inverted U-shaped EKC relationship between CO2 emissions and income by analyzing panel data of 83 countries for 1985-2013. Antweiler et al. (2001) and Coxhead (2003) illustrated this non-linear relationship between pollution and income from scale, composition, and technique spillover effects. Dinda (2004) summarized the possible explanations for EKC as: (1) an economy develops from a clean agrarian economy to a polluting industrial economy to a clean service economy and (2) people with higher incomes have higher preferences for environmental quality.

However, higher national incomes do not necessarily lead to greater efforts to contain the emission of pollutants. The empirical results of Holtz-Eakin and Selden (1995) and Shafik (1994) show that pollutant emissions monotonically increase with income levels. Magazzino and Cerulli (2019) used a responsiveness scores approach and panel data of Middle Eastern and North African (MENA) countries over the period 1971–2013 and found that the GDP per capita showed positive responsiveness scores on CO₂ emissions. The empirical results of Jaunky (2011), based on panel data of 36 high-income countries for the period 1980–2005, reveal that a 1% increase in GDP generated an increase of 0.68% in CO₂ emissions in the short term and 0.22% in the long term, which is inconsistent with EKC. The evidence provided by Halkos and Tsionas (2001) indicates a monotonic relationship between environmental degradation and income, and thus, rejects the EKC hypothesis.

⁴ With the recent deterioration of the "demographic dividend" in China, the advantage of labor costs no longer exists. Developed countries have moved their businesses to Southeast Asian countries that have lower labor costs. This compelled Chinese enterprises to retreat from the GVCs (Song and Wang, 2017).

Another strand of research unveiled an N-shaped rather than an inverted U-shaped relationship between pollution and output (e.g., Moomaw and Unruh, 1997; De Bruyn et al., 1998; Galeotti and Lanza, 1999; Millimet et al., 2003; Wang et al., 2019). Friedl and Getzner (2003) found a cubic relationship between GDP and CO₂ emissions for Austria for the period 1960–1999. Zheng et al. (2014) reiterated the cubic relationship by analyzing a panel data set of 30 provincial units in China from 1998 to 2010.

Accordingly, we propose the following hypothesis:

H3a. : GDP can significantly affect CO_2 emissions and the effect varies by country.

Magazzino (2016c) pointed out that the Toda and Yamamoto's Granger non-causality test (Toda and Yamamoto, 1995) showed a bidirectional causality between CO₂ emissions and economic growth in Italy over the period 1970–2006. Dinda and Coondoo (2006) employed bi-variate analysis and found bidirectional causality between CO₂ emissions and income for North America. Wang et al. (2016) established that increased CO₂ emissions in China during the period 1990–2012 led to augmented economic growth, though the impact was marginal. However, Magazzino (2016a) found that an increase in CO₂ emissions had a detrimental effect on the real GDP by studying the data of the South Caucasus area and Turkey from 1992 to 2013. Magazzino (2016b) explored the nexus between CO2 emissions, economic growth, and energy use for ten Middle Eastern countries during 1971-2006, using a PVAR technique. For the six GCC (Gulf Cooperation Council) countries, the response of economic growth (real GDP) to CO₂ emissions was negative in the estimated coefficients and impulse responses. For the other four non-GCC countries, CO2 emissions appeared to not have had any impacts on growth. Chontanawat (2020) found no evidence of causality running from CO2 emissions to GDP in ASEAN for the period 1971-2015.

Admittedly, the mixed nature of these empirical findings can be attributed to the econometric methods, research periods, and variables selected by researchers. However, we believe that it is essentially due to two effects. (1) Production effect. When CO₂ emissions are relatively low, some countries, especially low-income ones, will sacrifice the environment to promote production. They may even introduce relatively lower environmental standards in striving for opportunities to expand production. In this case, as CO₂ emissions increase, GDP rises as well. (2) Environmental protection effect. When CO₂ emissions reach a certain level, measures must be taken to ensure sustainable growth. Countries, especially high-income ones, will reduce or shut down high-emission production. Under these circumstances, as CO2 emissions grow, GDP may drop. In the early decades following the reform and opening policy in 1978, China emphasized the extreme importance of economic development (Liu et al., 2018). The extensive production mode with high energy consumption and high pollution promoted China to become the "world factory", which stimulated the rapid growth of China's economy. At that time, the production effect played a dominant role. Given resource depletion and environmental degradation, the focus of economic development has shifted into quality in recent decades. In 2012, China entered the state of "new normal" (Song and Wang, 2017), in which the economic growth rate is slowing and more attention is paid to the environment. At this stage, the environmental protection effect dominates.

Based on the discussion above, we propose:

H3b. : CO_2 emissions have a reverse impact on GDP and the effect varies by country.

Based on the literature review above, we found that previous research (1) ignored the potential reverse impacts of CO_2 emissions on GVC participation, GDP on GVC participation, and CO_2 emissions on GDP; (2) did not disentangle the relationships between GVC participation, CO_2 emissions, and GDP; and (3) did not pay sufficient attention to analyzing the way GVC participation in different sectors affects these

relationships or ascertaining the heterogeneity of these relationships between different income groups. To fill these gaps, this study used the PVAR model to analyze the dynamic relationships between GVC participation (in different sectors), CO₂ emissions, and GDP with the full, high-income, and low-income samples and explored the underlying mechanisms.

3. Model and methods

3.1. Model specification

The PVAR model was first proposed in Holtz-Eakin et al. (1988). It is a combination of the VAR model and panel data. The VAR model (Sims, 1980) treats all variables in the system as endogenous, which facilitates studies of bilateral or multi-lateral time-series causality between variables. PVAR models have been used in multiple applications across various fields of research (Love and Zicchino, 2006; Jawadi et al., 2016; Antonakakis et al., 2017; Magazzino, 2017).

To ensure environmentally-friendly economic growth, all countries and regions worldwide need to jointly tackle the problem of CO_2 emissions. Thus, we applied panel data in this study, from which individual heterogeneity can be observed. The PVAR model is superior to other methods because it follows the VAR model of treating all variables in the system as endogenous, which facilitates analyzing the dynamic relationships between them, while also allowing for unobserved individual heterogeneity by introducing the VAR in panel data settings, which is conducive to obtaining efficient estimates.

We followed a similar strategy of Abrigo and Love (2016) and Magazzino (2016a,b, 2017) and specified the model as:

$$Y_{it} = \sum_{j=1}^{p} A_j Y_{it-j} + u_i + e_{it} \ (i = 1, \dots, N; t = 1, \dots, T),$$
(1)

where Y_{it} represents a three-variable vector {GVC, LNGDPPC, LNCO2PC}. GVC, LNGDPPC, and LNCO2PC represent GVC participation degree, the logarithm of GDP per capita, and the logarithm of per capita CO₂ emissions, respectively. Let *i* index the cross-sectional observations and *t* the time period, respectively. In this study, N = 63 and T = 11. Y_{it-j} represents a *j*-period lag term of Y_{it} . The (3×3) matrices A_j are the parameters to be estimated. u_i and e_{it} are (1×3) vectors of dependent variable-specific panel fixed-effects and idiosyncratic errors.

We also examined the relationships between GVC participation in different industries, CO_2 emissions, and economic growth. We selected 12 industries and divided them into four industry groups according to their characteristics (see Section 3.2 for details). Thus, we had four additional regressions based on Model (1). Further, we analyzed the heterogeneous relationships between GVC, LNCO2PC, and LNGDPPC in high- and low-income countries, and thus we had two more regressions based on Model (1).

There is a restriction on applying the VAR procedure to panel data; that is, the underlying structure is the same for each cross-sectional unit (Love and Zicchino, 2006). To overcome the restriction on parameters, we allowed for "individual heterogeneity" by introducing fixed effects, denoted by u_i in Model (1).

As the fixed effects are correlated with the regressors due to the presence of lagged dependent variables in the right-hand side of the system of equations (Love and Zicchino, 2006), estimates would be biased even with a large N (Nickell, 1981). Theoretically, the bias approaches zero as T gets larger. However, Judson and Owen (1999) found significant bias in their research even when T = 30.

To further understand the dynamic relationships between our variables of interest, we presented impulse–response functions (IRFs) and forecast-error variance decompositions (FEVDs) based on the PVAR estimates. An IRF describes the reaction of one variable to a shock in another variable in the system. Referring to Lütkepohl (2007), we applied orthogonal impulse response functions (OIRFs) based on the Cholesky decomposition. The identifying assumption is that the variables that appear earlier in the system are more exogenous and the ones that appear later are more endogenous. In other words, the variables that come earlier in the ordering affect the variables that follow.⁵ The literature review in Section 2 has demonstrated the current period impacts of GVC participation on CO₂ emissions, GVC participation on GDP, and GDP on CO₂ emissions. Therefore, we reasonably assumed the order of the three variables from relatively exogenous to relatively endogenous as GVC, LNGDPPC, and LNCO2PC.⁶ Besides the OIRFs that present the year-by-year impulse–response dynamics, we also presented the cumulative impulse–response dynamics.

We then calculated the FEVDs to assess the importance of each variable in explaining the other variables. A FEVD shows the percentage of the variation in one variable that is explained by the shock in another variable (Abrigo and Love, 2016). We used the identification scheme employed in calculating the IRFs.

3.2. Variables and data

We collected data on per capita CO_2 emissions (CO2PC) and per capita GDP (GDPPC), expressed in constant 2010 USD, for 63 countries and regions from 2005 to 2015 from the World Bank database. The GVC participation degree (GVC) related data were obtained from the Trade in Value Added Database (2018 edition), developed by the Organization for Economic Co-operation and Development (OECD) and World Trade Organization (WTO).

3.2.1. GVC participation

The GVC participation degree consists of two parts: forward GVC participation degree (FGVC) and backward GVC participation degree (BGVC). Koopman et al. (2014) divided gross exports into final goods and intermediates and further divided intermediates into (1) finished and consumed goods in the importing country (FCIC); (2) goods that are processed and exported back to the exporting country (PEEC); and (3) goods processed and exported to a third country (PETC). Each part (or subpart) can be decomposed into domestic value added (DVA) and foreign value added (FVA) (see Fig. 4).

Referring to Koopman et al. (2014), and Wang et al. (2019), the FGVC, BGVC, and GVC participation degrees of country *i* at time *t* can be defined by Formulas (2)-(4), respectively.

specialization across countries drives the formation of GVCs and lowers the thresholds to international markets. Countries, especially developing and less developed ones, obtain more development opportunities through participating in GVCs. It will expand their production scales (i. e., the scale effect) but also enhance energy consumption, thus stimulating CO2 emissions. Second, GVC participation will promote production but reduce CO₂ emissions because of the technique spillover effect. Countries that actively and deeply participate in GVCs are more likely to benefit from the spillover effect of advanced technologies of trade partners. Advanced technologies enable countries to undertake cleaner production, which stimulates GDP and reduces CO2 emissions. Third, the composition effect is related to the economic structure driven by comparative advantages. To some extent, higher GVC participation degrees indicate deeper and wider intra-industry specialization, thus making the proportion of environmentally unfriendly production greater in countries that have comparative advantages in polluting products, while making countries whose comparative advantages lie in environmentally-friendly products more specialized in clean production.

Fourth, countries with high GVC participation degrees are relatively dependent on international markets and are likely to face fierce international competition. If a country enhances its international competitiveness through lower environmental standards, its polluting production may expand, and thus, both GDP and CO₂ emissions will increase. However, if a country gains competitiveness via R&D and innovation, its GDP will grow and emissions will decrease. Thus, the impact of GVC participation will be positive on GDP but ambiguous on CO₂ emissions due to the competition effect. Moreover, as an economy develops, people's preference for a high-quality environment grows. Environmentally-friendly production will finally survive in international competition. Therefore, we believe that the competition effect is conducive to promoting production while reducing emissions in the long run. In sum, the net impacts of GVC participation on CO₂ emissions and GDP are ambiguous.

We can obtain the GVC participation degree for an industry by replacing all the variables in (2) to (4) with industry-level values. To further analyze the relationships between GVC participation, CO_2 emissions, and economic growth, we collected data on GVC participation degrees of 12 industries from the Trade in Value Added Database (2018 edition) developed by OECD and WTO. We divided the industries

Forward Particip. Degree in
$$GVC_{it} = \frac{DVA \ embodied \ in \ foreign \ exports_{it}}{Gross \ exports_{it}} = \frac{DVA_{PETC \ it}}{Gross \ exports_{it}},$$
 (2)

Backward Particip. Degree in
$$GVC_{it} = \frac{FVA_{it}}{Gross \ exports_{it}} = \frac{\left(FVA_{Final \ goods} + FVA_{FCIC} + FVA_{PEEC} + FVA_{PETC}\right)_{it}}{Gross \ exports_{it}},$$
(3)

$$Particip. Degree in GVC_{it} = FGVC_{it} + BGVC_{it} = \frac{DVA_{PETCit} + (FVA_{Final \ goods} + FVA_{FCIC} + FVA_{PEEC} + FVA_{PETC})_{it}}{Gross \ exports_{it}}.$$
(4)

The GVC–CO₂ and GVC–GDP relationships are determined by four effects—scale, technique spillover, composition, and competition effects. First, GVC participation will stimulate GDP growth and CO₂ emissions through the scale effect. The in-depth intra-industry

into four groups based on their contributions to domestic CO_2 emissions and value added, following three steps: First, we obtained data on CO_2 emissions by industry from the Air and Climate Database (2018 edition) and value added by industry from the Trade in Value-Added Database (2018 edition), respectively, developed by the OECD and the WTO. By matching the two databases, we obtained measures of CO_2 emissions and value added for 12 industries in 32 out of 63 countries and regions from 2012 to 2015. Second, we computed the mean of CO_2 emissions as well as the mean of value added of these 32 countries from 2012 to 2015 by industry. A scatter plot of selected industries is shown in Fig. 5. Third, we split Fig. 5 into four quadrants based on the median values of CO_2 emissions (vertical axis) and value added (horizontal axis). Industries

⁵ See Hamilton (1994) and Abrigo and Love (2016) for the derivation and discussion of IRFs.

⁶ The VAR model in this study under these assumptions is similar to the structural VAR (SVAR) model with zero short-run restrictions, also known as Cholesky, recursive, or orthogonal identifications (Stock and Watson, 2001).



Fig. 4. The decomposition of gross exports.

Note: DVA and FVA stand for domestic value added and foreign value added, respectively.



Fig. 5. Scatter plot of the mean of CO_2 emissions (vertical axis) and mean of value added (horizontal axis) of 12 industries in 32 countries from 2012 to 2015.

Data Sources: CO_2 emissions data by industry were collected from the Air and Climate Database (2018 edition) and value-added data by industry were obtained from the Trade in Value Added Database (2018 edition), respectively, developed by the Organization for Economic Co-operation and Development and World Trade Organization.

Table 1

Industry	groups.
----------	---------

No.	Industry	Group
3	Manufacturing	
4	Construction	
5	Wholesale and retail trade; repair of motor vehicles	нн
6	Transportation and storage	
1	Agriculture, forestry, and fishing	
2	Mining and quarrying	пL
7	Accommodation and food services	
8	Information and communication	
9	Financial and insurance activities	LL
11	Education	
10	Real estate activities	
12	Human health and social work	LFI

that fall in the first quadrant were classified into the"HH" group, that is, the high-CO₂ emissions and high value-added group. Similarly, industries falling in the second, third, and fourth quadrants were classified into the "HL," "LL," and "LH" groups, respectively. See Table 1 for details of the industry groups.⁷

Owing to diverse industry characteristics, we expected heterogeneous $\text{GVC}-\text{CO}_2$ and GVC-GDP relationships by industry group. Theoretically, the response of CO_2 emissions to GVC participation was expected to be stronger in high- CO_2 industry groups, and that of GDP to GVC participation was expected to be stronger in high value-added industry groups. However, since the aforementioned four effects work simultaneously, the expected empirical results would be ambiguous.

 $^{^{7}}$ We refer to these four groups as "HH," "HL," "LH," and "LL," but we remind readers that this distinction is relative and is based on the median level of the mean values of CO₂ emissions and value added by industry across 32 countries from 2012 to 2015.

3.2.2. Per capita CO₂ emissions

Although previous research has hardly studied the effects of CO_2 emissions on GVC participation and GDP, we argued that CO_2 emissions will affect GVC participation because if participating in GVCs helps reduce CO_2 emissions, holding other factors constant, countries will tend to enhance their GVC participation degrees and vice versa. The CO_2 -GDP relationship is more complicated. If CO_2 emissions decrease as the economy develops, countries will embrace economic growth. However, if economic development promotes CO_2 emissions, a country may or may not sacrifice economic growth to protect the environment; it depends on its level of economic development and public environmental awareness.

3.2.3. Per capita GDP

The impact of GDP on CO_2 emissions has been analyzed extensively (see Section 2 for details). However, research on the effect of GDP on GVC participation is rare. We proposed that this effect is ambiguous because, on the one hand, higher GDP indicates higher productivity and greater international influence. A country with high GDP is more capable of participating in GVCs. On the other hand, countries that actively and deeply participate in GVCs are more likely to be affected by shocks in international markets. If there are serious adverse shocks in the international economy, it is the countries with high GDP that are more likely to cut their linkages with GVCs because they are more capable of being self-sufficient.

In addition, we divided 63 countries and regions into two groups: the high-income group includes 39 high-income ones, and the low-income group comprises the rest.⁸ Considering the heterogeneity of countries in terms of natural resource endowments, comparative advantages, production styles, environmental protection awareness, and responses to international shocks, we expect that the aforementioned dynamic relationships vary across countries with different incomes.

Table 2 reports the summary statistics of all variables. Single country (region) statistics are available upon request. The mean values of all variables are positive. LNGDPPC and LNCO2PC have negative values of skewness, indicating that the distribution is left-skewed, with more observations on the right tail. However, the variables show similar values for the mean and median, indicating that a normal distribution emerges. Moreover, 10-trim values are near to the mean, and standard deviation to the pseudo standard deviation (PSD), which are in line with the fact that the inter-quartile range (IQR) shows the absence of outliers in our sample. The correlation matrix indicates that our series are significantly correlated at the 1% level (Table 3).

4. Empirical results

4.1. Diagnostic tests

Before applying the PVAR approach, we carried out a series of panel unit root tests including Levin et al. (2002), Harris and Tzavalis (1999), Breitung (2000), Im et al. (2003), Fisher-type (Choi, 2001), and Pesaran (2007). All tests except the CIPS test confirm that our variables are (trend) stationary. The CIPS test of some variables failed to reject the presence of cross-section dependence, including at the first-difference level. We applied Pesaran's (2004) cross-sectional dependence (CSD) test, which also failed to reject the presence of cross-sectional dependence in some cases. Although the spatial issue is beyond the scope of this research, we addressed the issue in the robustness check section.

Further, the results of the Dumitrescu-Hurling panel causality tests (Dumitrescu and Hurlin, 2012) reveal the bidirectional causality

between each pair of variables of primary interest. The optimal lag chosen for the PVAR model is based on balancing the following aspects: the model has a small value of the Modified Akaike Information Criterion (MAIC), Modified Bayesian Information Criterion (MBIC), or Modified Hannan-Quinn Information Criterion (MQIC), and passes the Hansen's (1982) over-identifying test. Based on these selection criteria, we specified a 1-period lag order in the baseline regressions and the regressions grouped by income, and a 2-period lag order in the regressions containing GVC participation in different industry groups.⁹

4.2. PVAR analysis

4.2.1. Baseline regression results

Table 4 presents the results of the PVAR(1) model which considers the overall GVC participation. We found the following: (1) All three variables are significantly and positively correlated with their past values, which is in line with their trends. (2) The 1-period lag of GVC has significantly negative impacts on LNCO2PC and LNGDPPC. As GVC increases by 1%, CO2PC will decrease by 0.012%, more than the reduction in GDPPC (0.007%). (3) The 1-period lag of LNGDPPC will significantly and negatively affect GVC. The GVC will decrease by 6.4% when GDPPC increases by 1%. The impact of LNGDPPC on LNCO2PC one year ahead is positive but insignificant. (4) The 1-period lag of LNCO2PC will significantly and negatively influence LNGDPPC, while positively but insignificantly affecting GVC. As per capita CO_2 emissions increase by 1%, the GDP per capita will decrease by 0.18%, and vice versa.

4.2.2. Results by industry group

Table 5 presents the major PVAR(2) regression results when GVC participation in different industry groups is considered.¹⁰ Panels I–IV show the results of HH, HL, LL, and LH groups, respectively. We found the following: (1) The 1-period lag of GVC participation in HH and HL industry groups will significantly and negatively affect LNCO2PC, while the coefficients are negative but insignificant in LL and LH. Moreover, the 1-period lag of GVC participation in HH, and LL groups is significantly negatively correlated with LNGDPPC, and the coefficient of LL is smaller, while the coefficient of the HL group is insignificant. Furthermore, the coefficients are insignificant when the 2-period lag values of GVC participation are considered.

(2) The 1-period lag of LNGDPPC is positively correlated with GVC participation in all groups, while the 2-period lag of LNGDPPC is negatively correlated, although the coefficients are insignificant in the HL and LH groups. (3) The 1-period lag of LNCO2PC is positively correlated with GVC participation in the HH, HL, and LH groups, while the 2-period lag of LNCO2PC is negatively correlated, although the coefficients are insignificant in the HH group. The 1-period lag of LNCO2PC is significantly negatively correlated with GVC participation in the LL group and the coefficient of the 2-period lag is negative but insignificant.

4.2.3. Results by income group

Table 6 presents the PVAR(1) regression results when countries with different incomes are considered. We found the following: (1) The 1-period lag of GVC shows negative impacts on LNGDPPC and LNCO2PC in both groups, while the coefficients of the low-income group are insignificant and smaller. (2) The 1-period lag of LNGDPPC is positively but insignificantly correlated with GVC in the high-income group, while significantly negatively correlated with GVC in the low-income group.

⁸ Countries and regions, whose average income during 2005 to 2015 was above the average of high-income standards announced by the World Bank over the same time period, were classified as the high-income group, while other countries and regions were classified as the low-income group.

 $^{^{9}}$ The results of unit root tests, causality tests, the MAIC, MBIC, and MQIC values are available upon request.

¹⁰ As far as industries are considered, we only showed the results of our main focus, i.e., the impacts of GVC participation in different industry groups on LNGDPPC and LNCO2PC, and the reverse impacts of LNGDPPC and LNCO2PC on GVC participation in different industry groups.

Table 2

Descriptive statistics.

-															
Variable	Obs.	Mean	Median	SD	Variance	Min	Max	Skewness	Kurtosis	IQR	Range	CV	SE	10-Trim	PSD
GVC	693	45.369	44.3	9.25	85.561	23.5	79.4	0.615	3.865	10.5	55.9	0.204	0.351	44.884	7.778
GVC_HH	693	25.719	24.625	8.122	65.964	8.2	50.075	0.591	2.985	10.9	41.875	0.316	0.309	25.185	8.074
GVC_HL	693	18.763	17.55	8.77	76.915	3.15	52.15	0.982	4.166	10.6	49	0.467	0.333	17.880	7.852
GVC_LL	693	12.481	10.675	7.389	54.597	3.225	46.25	2.21	9.06	6.95	43.025	0.592	0.281	11.305	5.148
GVC_LH	693	9.007	8.05	4.399	19.353	2.25	25.1	0.867	3.281	6.1	22.85	0.488	0.167	8.551	4.519
LNGDPPC	693	9.733	9.91	1.089	1.187	6.419	11.626	-0.732	3.051	1.575	5.207	0.112	0.041	9.837	1.167
LNCO2PC	693	1.741	1.874	0.803	0.644	-1.565	3.212	-1.031	4.527	0.805	4.777	0.461	0.031	1.806	0.597
By income grou	D														
GVC_H	429	46.73	44.5	9.779	95.634	25.5	79.4	0.771	3.624	10.500	53.9	0.209	0.472	46.014	7.778
LNGDPPC_H	429	10.428	10.518	0.529	0.279	9.262	11.626	-0.181	2.442	0.785	2.364	0.051	0.026	10.435	0.581
LNCO2PC_H	429	2.109	2.079	0.453	0.205	1.21	3.212	0.383	2.564	0.572	2.002	0.215	0.022	2.089	0.424
GVC_L	264	43.158	44.15	7.846	61.563	23.5	61	-0.233	2.537	10.250	37.5	0.182	0.483	43.36	7.593
LNGDPPC_L	264	8.604	8.883	0.775	0.6	6.419	9.597	-0.995	3.039	1.041	3.178	0.090	0.048	8.703	0.771
LNCO2PC_L	264	1.142	1.338	0.882	0.779	-1.565	2.75	-0.431	3.073	1.322	4.315	0.772	0.054	1.174	0.979
	Variable GVC GVC_HH GVC_HL GVC_LL GVC_LL LNCO2PC LNCO2PC H LNCO2PC_H LNCO2PC_H LNCO2PC_H LNCO2PC_L LNCO2PC_L LNCO2PC_L	Variable Obs. GVC 693 GVC_HL 693 GVC_HL 693 GVC_LL 693 GVC_LL 693 LNGDPPC 693 LNCO2PC 693 By income group GVC_H GVC_H 429 LNGDPPC_H 429 GVC_L 264 LNGDPPC_L 264	Variable Obs. Mean GVC 693 45.369 GVC,HL 693 25.719 GVC,HL 693 12.481 GVC,LL 693 12.481 GVC,LH 693 9.007 LNGDPPC 693 9.733 LNCO2PC 693 1.741 By income group GVC_H 429 GVC_H 429 46.73 LNGDPPC_H 429 10.428 LNCO2PC,H 429 2.109 GVC_L 264 8.604 LNCO2PC L 264 1.142	Variable Obs. Mean Median GVC 693 45.369 44.3 GVC_HH 693 25.719 24.625 GVC_HL 693 18.763 17.55 GVC_LL 693 12.481 10.675 GVC_LH 693 9.733 9.91 LNCO2PC 693 1.741 1.874 By income group GVC_H 429 46.73 44.5 LNGDPPC_H 429 10.428 10.518 10.518 LNGDPPC_H 429 2.109 2.079 GVC_L GVC_L 264 43.158 44.15 1.NGDPPC_L LNGDPPC_L 264 8.604 8.883 1.NCO2PC L 264 1.142 1.338	Variable Obs. Mean Median SD GVC 693 45.369 44.3 9.25 GVC_HH 693 25.719 24.625 8.122 GVC_HL 693 18.763 17.55 8.77 GVC_IL 693 12.481 10.675 7.389 GVC_IL 693 9.007 8.05 4.399 LNGDPPC 693 9.733 9.91 1.089 LNCO2PC 693 1.741 1.874 0.803 By income group GVC_H 429 46.73 44.5 9.779 LNGDPPC_H 429 10.428 10.518 0.529 LNCO2PC_H 429 2.109 2.079 0.453 GVC_L 264 8.604 8.883 0.775 LNGDPPC_L 264 8.604 8.883 0.775 LNCO2PC L 264 1.142 1.338 0.882	Variable Obs. Mean Median SD Variance GVC 693 45.369 44.3 9.25 85.561 GVC HH 693 25.719 24.625 8.122 65.964 GVC HL 693 18.763 17.55 8.77 76.915 GVC LL 693 12.481 10.675 7.389 54.597 GVC LH 693 9.007 8.05 4.399 19.353 LNGDPPC 693 1.741 1.874 0.803 0.644 By income group GVC_H 429 46.73 44.5 9.779 95.634 LNGDPPC H 429 10.428 10.518 0.529 0.279 LNCO2PC H 429 2.109 2.079 0.453 0.205 GVC L 264 43.158 44.15 7.846 61.563 LNGDPPC L 264 8.604 8.883 0.775 0.6 LNCO2PC L 264 1.142 1.338 0.882 </td <td>Variable Obs. Mean Median SD Variance Min GVC 693 45.369 44.3 9.25 85.561 23.5 GVC HH 693 25.719 24.625 8.122 65.964 8.2 GVC HL 693 18.763 17.55 8.77 76.915 3.12 GVC_LL 693 12.481 10.675 7.389 54.597 3.225 GVC_LH 693 9.007 8.05 4.399 19.353 2.25 LNGDPPC 693 9.733 9.91 1.089 1.187 6.419 LNCO2PC 693 1.741 1.874 0.803 0.644 -1.565 By income group GVC_H 429 46.73 44.5 9.779 95.634 25.5 LNGDPPC_H 429 10.428 10.518 0.529 0.279 9.262 LNCO2PC_H 429 2.109 2.079 0.453 0.205 1.21</td> <td>Variable Obs. Mean Median SD Variance Min Max GVC 693 45.369 44.3 9.25 85.561 23.5 79.4 GVC HI 693 25.719 24.625 8.122 65.964 8.2 50.075 GVC HL 693 18.763 17.55 8.77 76.915 3.15 52.15 GVC_LL 693 12.481 10.675 7.389 54.597 3.225 46.25 GVC_LH 693 9.007 8.05 4.399 19.353 2.25 25.1 LNGDPPC 693 9.733 9.91 1.089 1.187 6.419 11.626 LNCO2PC 693 1.741 1.874 0.803 0.644 -1.565 3.212 By income group GVC_H 429 46.73 44.5 9.779 95.634 25.5 79.4 LNGDPPC_H 429 10.428 10.518 0.529 0.279 <td< td=""><td>Variable Obs. Mean Median SD Variance Min Max Skewness GVC 693 45.369 44.3 9.25 85.561 23.5 79.4 0.615 GVC HH 693 25.719 24.625 8.122 65.964 8.2 50.075 0.591 GVC HL 693 18.763 17.55 8.77 76.915 3.15 52.15 0.982 GVC LL 693 12.481 10.675 7.389 54.597 3.225 46.25 2.21 GVC LH 693 9.007 8.05 4.399 19.353 2.25 25.1 0.867 LNGDPPC 693 9.733 9.91 1.089 1.187 6.419 11.626 -0.732 LNCO2PC 693 1.741 1.874 0.803 0.644 -1.565 3.212 -1.031 By income group GVC H 429 46.73 44.5 9.779 95.634 25.5 79.4 0.771<</td><td>Variable Obs. Mean Median SD Variance Min Max Skewness Kurtosis GVC 693 45.369 44.3 9.25 85.561 23.5 79.4 0.615 3.865 GVC HH 693 25.719 24.625 8.122 65.964 8.2 50.075 0.591 2.985 GVC HL 693 18.763 17.55 8.77 76.915 3.15 52.15 0.982 4.166 GVC LL 693 9.007 8.05 4.399 19.353 2.25 25.1 0.867 3.281 LNGDPPC 693 9.733 9.91 1.089 1.187 6.419 11.626 -0.732 3.051 LNCO2PC 693 1.741 1.874 0.803 0.644 -1.565 3.212 -1.031 4.527 By income group GVC H 429 46.73 44.5 9.779 95.634 25.5 79.4 0.771 3.624 LNGD</td><td>Variance Obs. Mean Median SD Variance Min Max Skewness Kurtosis IQR GVC 693 45.369 44.3 9.25 85.561 23.5 79.4 0.615 3.865 10.5 GVC HI 693 25.719 24.625 8.122 65.964 8.2 50.075 0.591 2.985 10.9 GVC HL 693 18.763 17.55 8.77 76.915 3.15 52.15 0.982 4.166 10.6 GVC_LL 693 12.481 10.675 7.389 54.597 3.225 46.25 2.21 9.06 6.95 GVC_LH 693 9.007 8.05 4.399 19.353 2.25 25.1 0.867 3.281 6.1 LNGDPPC 693 9.733 9.91 1.089 1.187 6.419 11.626 -0.732 3.051 1.575 LNCO2PC 693 1.741 1.874 0.803</td><td>Variance Min Max Skewness Kurtosis IQR Range GVC 693 45.369 44.3 9.25 85.561 23.5 79.4 0.615 3.865 10.5 55.9 GVC HH 693 25.719 24.625 8.122 65.964 8.2 50.075 0.591 2.985 10.9 41.875 GVC HL 693 18.763 17.55 8.77 76.915 3.15 52.15 0.982 41.66 10.6 49 GVC_LL 693 12.481 10.675 7.389 54.597 3.225 46.25 2.21 9.06 6.95 43.025 GVC_LH 693 9.007 8.05 4.399 19.353 2.25 25.1 0.867 3.281 6.1 22.85 LNGOPPC 693 9.733 9.91 1.089 1.187 6.419 11.626 -0.732 3.051 1.575 5.207 LNCO2PC 693 1.741<</td><td>Variable Obs. Mean Median SD Variance Min Max Skewness Kurtosis IQR Range CV GVC 693 45.369 44.3 9.25 85.561 23.5 79.4 0.615 3.865 10.5 55.9 0.204 GVC_HH 693 25.719 24.625 8.122 65.964 8.2 50.075 0.591 2.985 10.9 41.875 0.316 GVC_HL 693 18.763 17.55 8.77 76.915 3.15 52.15 0.982 4.166 10.6 49 0.467 GVC_LH 693 9.007 8.05 4.399 19.353 2.25 25.1 0.867 3.281 6.1 22.85 0.488 LNGDPPC 693 9.733 9.91 1.089 1.187 6.419 11.626 -0.732 3.051 1.575 5.207 0.112 LNCO2PC 693 1.741 1.874 0.803 0.644</td><td>Variable Obs. Mean Median SD Variance Min Max Skewness Kurtosis IQR Range CV SE GVC 693 45.369 44.3 9.25 85.561 23.5 79.4 0.615 3.865 10.5 55.9 0.204 0.351 GVC_HH 693 25.719 24.625 8.122 65.964 8.2 50.075 0.591 2.985 10.9 41.875 0.316 0.309 GVC_HL 693 18.763 17.55 8.77 76.915 3.15 52.15 0.982 4.166 10.6 49 0.467 0.333 GVC_LH 693 9.007 8.05 4.399 19.353 2.25 25.1 0.867 3.281 6.1 22.85 0.488 0.167 LNGDPPC 693 9.733 9.91 1.089 1.187 6.419 11.626 -0.732 3.051 1.575 5.207 0.112 0.041 <</td><td>Variable Obs. Mean Median SD Variance Min Max Skewness Kurtosis IQR Range CV SE 10-Trim GVC 693 45.369 44.3 9.25 85.561 23.5 79.4 0.615 3.865 10.5 55.9 0.204 0.351 44.884 GVC_HH 693 25.719 24.625 8.122 65.964 8.2 50.075 0.591 2.985 10.9 41.875 0.316 0.309 25.185 GVC_HL 693 18.763 17.55 8.77 76.915 3.15 52.15 0.982 4.166 10.6 49 0.467 0.333 17.880 GVC_LH 693 9.007 8.05 4.399 19.353 2.25 25.1 0.867 3.281 6.1 22.85 0.488 0.167 8.551 LNGDPPC 693 9.733 9.91 1.089 1.187 6.419 11.626 -0.732 3.051</td></td<></td>	Variable Obs. Mean Median SD Variance Min GVC 693 45.369 44.3 9.25 85.561 23.5 GVC HH 693 25.719 24.625 8.122 65.964 8.2 GVC HL 693 18.763 17.55 8.77 76.915 3.12 GVC_LL 693 12.481 10.675 7.389 54.597 3.225 GVC_LH 693 9.007 8.05 4.399 19.353 2.25 LNGDPPC 693 9.733 9.91 1.089 1.187 6.419 LNCO2PC 693 1.741 1.874 0.803 0.644 -1.565 By income group GVC_H 429 46.73 44.5 9.779 95.634 25.5 LNGDPPC_H 429 10.428 10.518 0.529 0.279 9.262 LNCO2PC_H 429 2.109 2.079 0.453 0.205 1.21	Variable Obs. Mean Median SD Variance Min Max GVC 693 45.369 44.3 9.25 85.561 23.5 79.4 GVC HI 693 25.719 24.625 8.122 65.964 8.2 50.075 GVC HL 693 18.763 17.55 8.77 76.915 3.15 52.15 GVC_LL 693 12.481 10.675 7.389 54.597 3.225 46.25 GVC_LH 693 9.007 8.05 4.399 19.353 2.25 25.1 LNGDPPC 693 9.733 9.91 1.089 1.187 6.419 11.626 LNCO2PC 693 1.741 1.874 0.803 0.644 -1.565 3.212 By income group GVC_H 429 46.73 44.5 9.779 95.634 25.5 79.4 LNGDPPC_H 429 10.428 10.518 0.529 0.279 <td< td=""><td>Variable Obs. Mean Median SD Variance Min Max Skewness GVC 693 45.369 44.3 9.25 85.561 23.5 79.4 0.615 GVC HH 693 25.719 24.625 8.122 65.964 8.2 50.075 0.591 GVC HL 693 18.763 17.55 8.77 76.915 3.15 52.15 0.982 GVC LL 693 12.481 10.675 7.389 54.597 3.225 46.25 2.21 GVC LH 693 9.007 8.05 4.399 19.353 2.25 25.1 0.867 LNGDPPC 693 9.733 9.91 1.089 1.187 6.419 11.626 -0.732 LNCO2PC 693 1.741 1.874 0.803 0.644 -1.565 3.212 -1.031 By income group GVC H 429 46.73 44.5 9.779 95.634 25.5 79.4 0.771<</td><td>Variable Obs. Mean Median SD Variance Min Max Skewness Kurtosis GVC 693 45.369 44.3 9.25 85.561 23.5 79.4 0.615 3.865 GVC HH 693 25.719 24.625 8.122 65.964 8.2 50.075 0.591 2.985 GVC HL 693 18.763 17.55 8.77 76.915 3.15 52.15 0.982 4.166 GVC LL 693 9.007 8.05 4.399 19.353 2.25 25.1 0.867 3.281 LNGDPPC 693 9.733 9.91 1.089 1.187 6.419 11.626 -0.732 3.051 LNCO2PC 693 1.741 1.874 0.803 0.644 -1.565 3.212 -1.031 4.527 By income group GVC H 429 46.73 44.5 9.779 95.634 25.5 79.4 0.771 3.624 LNGD</td><td>Variance Obs. Mean Median SD Variance Min Max Skewness Kurtosis IQR GVC 693 45.369 44.3 9.25 85.561 23.5 79.4 0.615 3.865 10.5 GVC HI 693 25.719 24.625 8.122 65.964 8.2 50.075 0.591 2.985 10.9 GVC HL 693 18.763 17.55 8.77 76.915 3.15 52.15 0.982 4.166 10.6 GVC_LL 693 12.481 10.675 7.389 54.597 3.225 46.25 2.21 9.06 6.95 GVC_LH 693 9.007 8.05 4.399 19.353 2.25 25.1 0.867 3.281 6.1 LNGDPPC 693 9.733 9.91 1.089 1.187 6.419 11.626 -0.732 3.051 1.575 LNCO2PC 693 1.741 1.874 0.803</td><td>Variance Min Max Skewness Kurtosis IQR Range GVC 693 45.369 44.3 9.25 85.561 23.5 79.4 0.615 3.865 10.5 55.9 GVC HH 693 25.719 24.625 8.122 65.964 8.2 50.075 0.591 2.985 10.9 41.875 GVC HL 693 18.763 17.55 8.77 76.915 3.15 52.15 0.982 41.66 10.6 49 GVC_LL 693 12.481 10.675 7.389 54.597 3.225 46.25 2.21 9.06 6.95 43.025 GVC_LH 693 9.007 8.05 4.399 19.353 2.25 25.1 0.867 3.281 6.1 22.85 LNGOPPC 693 9.733 9.91 1.089 1.187 6.419 11.626 -0.732 3.051 1.575 5.207 LNCO2PC 693 1.741<</td><td>Variable Obs. Mean Median SD Variance Min Max Skewness Kurtosis IQR Range CV GVC 693 45.369 44.3 9.25 85.561 23.5 79.4 0.615 3.865 10.5 55.9 0.204 GVC_HH 693 25.719 24.625 8.122 65.964 8.2 50.075 0.591 2.985 10.9 41.875 0.316 GVC_HL 693 18.763 17.55 8.77 76.915 3.15 52.15 0.982 4.166 10.6 49 0.467 GVC_LH 693 9.007 8.05 4.399 19.353 2.25 25.1 0.867 3.281 6.1 22.85 0.488 LNGDPPC 693 9.733 9.91 1.089 1.187 6.419 11.626 -0.732 3.051 1.575 5.207 0.112 LNCO2PC 693 1.741 1.874 0.803 0.644</td><td>Variable Obs. Mean Median SD Variance Min Max Skewness Kurtosis IQR Range CV SE GVC 693 45.369 44.3 9.25 85.561 23.5 79.4 0.615 3.865 10.5 55.9 0.204 0.351 GVC_HH 693 25.719 24.625 8.122 65.964 8.2 50.075 0.591 2.985 10.9 41.875 0.316 0.309 GVC_HL 693 18.763 17.55 8.77 76.915 3.15 52.15 0.982 4.166 10.6 49 0.467 0.333 GVC_LH 693 9.007 8.05 4.399 19.353 2.25 25.1 0.867 3.281 6.1 22.85 0.488 0.167 LNGDPPC 693 9.733 9.91 1.089 1.187 6.419 11.626 -0.732 3.051 1.575 5.207 0.112 0.041 <</td><td>Variable Obs. Mean Median SD Variance Min Max Skewness Kurtosis IQR Range CV SE 10-Trim GVC 693 45.369 44.3 9.25 85.561 23.5 79.4 0.615 3.865 10.5 55.9 0.204 0.351 44.884 GVC_HH 693 25.719 24.625 8.122 65.964 8.2 50.075 0.591 2.985 10.9 41.875 0.316 0.309 25.185 GVC_HL 693 18.763 17.55 8.77 76.915 3.15 52.15 0.982 4.166 10.6 49 0.467 0.333 17.880 GVC_LH 693 9.007 8.05 4.399 19.353 2.25 25.1 0.867 3.281 6.1 22.85 0.488 0.167 8.551 LNGDPPC 693 9.733 9.91 1.089 1.187 6.419 11.626 -0.732 3.051</td></td<>	Variable Obs. Mean Median SD Variance Min Max Skewness GVC 693 45.369 44.3 9.25 85.561 23.5 79.4 0.615 GVC HH 693 25.719 24.625 8.122 65.964 8.2 50.075 0.591 GVC HL 693 18.763 17.55 8.77 76.915 3.15 52.15 0.982 GVC LL 693 12.481 10.675 7.389 54.597 3.225 46.25 2.21 GVC LH 693 9.007 8.05 4.399 19.353 2.25 25.1 0.867 LNGDPPC 693 9.733 9.91 1.089 1.187 6.419 11.626 -0.732 LNCO2PC 693 1.741 1.874 0.803 0.644 -1.565 3.212 -1.031 By income group GVC H 429 46.73 44.5 9.779 95.634 25.5 79.4 0.771<	Variable Obs. Mean Median SD Variance Min Max Skewness Kurtosis GVC 693 45.369 44.3 9.25 85.561 23.5 79.4 0.615 3.865 GVC HH 693 25.719 24.625 8.122 65.964 8.2 50.075 0.591 2.985 GVC HL 693 18.763 17.55 8.77 76.915 3.15 52.15 0.982 4.166 GVC LL 693 9.007 8.05 4.399 19.353 2.25 25.1 0.867 3.281 LNGDPPC 693 9.733 9.91 1.089 1.187 6.419 11.626 -0.732 3.051 LNCO2PC 693 1.741 1.874 0.803 0.644 -1.565 3.212 -1.031 4.527 By income group GVC H 429 46.73 44.5 9.779 95.634 25.5 79.4 0.771 3.624 LNGD	Variance Obs. Mean Median SD Variance Min Max Skewness Kurtosis IQR GVC 693 45.369 44.3 9.25 85.561 23.5 79.4 0.615 3.865 10.5 GVC HI 693 25.719 24.625 8.122 65.964 8.2 50.075 0.591 2.985 10.9 GVC HL 693 18.763 17.55 8.77 76.915 3.15 52.15 0.982 4.166 10.6 GVC_LL 693 12.481 10.675 7.389 54.597 3.225 46.25 2.21 9.06 6.95 GVC_LH 693 9.007 8.05 4.399 19.353 2.25 25.1 0.867 3.281 6.1 LNGDPPC 693 9.733 9.91 1.089 1.187 6.419 11.626 -0.732 3.051 1.575 LNCO2PC 693 1.741 1.874 0.803	Variance Min Max Skewness Kurtosis IQR Range GVC 693 45.369 44.3 9.25 85.561 23.5 79.4 0.615 3.865 10.5 55.9 GVC HH 693 25.719 24.625 8.122 65.964 8.2 50.075 0.591 2.985 10.9 41.875 GVC HL 693 18.763 17.55 8.77 76.915 3.15 52.15 0.982 41.66 10.6 49 GVC_LL 693 12.481 10.675 7.389 54.597 3.225 46.25 2.21 9.06 6.95 43.025 GVC_LH 693 9.007 8.05 4.399 19.353 2.25 25.1 0.867 3.281 6.1 22.85 LNGOPPC 693 9.733 9.91 1.089 1.187 6.419 11.626 -0.732 3.051 1.575 5.207 LNCO2PC 693 1.741<	Variable Obs. Mean Median SD Variance Min Max Skewness Kurtosis IQR Range CV GVC 693 45.369 44.3 9.25 85.561 23.5 79.4 0.615 3.865 10.5 55.9 0.204 GVC_HH 693 25.719 24.625 8.122 65.964 8.2 50.075 0.591 2.985 10.9 41.875 0.316 GVC_HL 693 18.763 17.55 8.77 76.915 3.15 52.15 0.982 4.166 10.6 49 0.467 GVC_LH 693 9.007 8.05 4.399 19.353 2.25 25.1 0.867 3.281 6.1 22.85 0.488 LNGDPPC 693 9.733 9.91 1.089 1.187 6.419 11.626 -0.732 3.051 1.575 5.207 0.112 LNCO2PC 693 1.741 1.874 0.803 0.644	Variable Obs. Mean Median SD Variance Min Max Skewness Kurtosis IQR Range CV SE GVC 693 45.369 44.3 9.25 85.561 23.5 79.4 0.615 3.865 10.5 55.9 0.204 0.351 GVC_HH 693 25.719 24.625 8.122 65.964 8.2 50.075 0.591 2.985 10.9 41.875 0.316 0.309 GVC_HL 693 18.763 17.55 8.77 76.915 3.15 52.15 0.982 4.166 10.6 49 0.467 0.333 GVC_LH 693 9.007 8.05 4.399 19.353 2.25 25.1 0.867 3.281 6.1 22.85 0.488 0.167 LNGDPPC 693 9.733 9.91 1.089 1.187 6.419 11.626 -0.732 3.051 1.575 5.207 0.112 0.041 <	Variable Obs. Mean Median SD Variance Min Max Skewness Kurtosis IQR Range CV SE 10-Trim GVC 693 45.369 44.3 9.25 85.561 23.5 79.4 0.615 3.865 10.5 55.9 0.204 0.351 44.884 GVC_HH 693 25.719 24.625 8.122 65.964 8.2 50.075 0.591 2.985 10.9 41.875 0.316 0.309 25.185 GVC_HL 693 18.763 17.55 8.77 76.915 3.15 52.15 0.982 4.166 10.6 49 0.467 0.333 17.880 GVC_LH 693 9.007 8.05 4.399 19.353 2.25 25.1 0.867 3.281 6.1 22.85 0.488 0.167 8.551 LNGDPPC 693 9.733 9.91 1.089 1.187 6.419 11.626 -0.732 3.051

Notes: 1. GVC, GVC_HH, GVC_HL, GVC_LL, GVC_LH, GVC_L, and GVC_L stand for the overall global value chain participation degree and degrees in the HH, HL, LL, and LH industry groups as well as high (H) and low(L) income groups, respectively. 2. LNGDPPC, LNGDPPC_H, and LNGDPPC_L are the logarithm of GDP per capita in the full, high-income, and low-income sample, respectively. 3. LNCO2PC, LNCO2PC_H, and LNCO2PC_L stand for the logarithm of per capita CO₂ emission in the full, high-income, and low-income sample, respectively. 4. As far as industry groups are concerned, we focus on the dynamic relationships between GVC participation in each group, the overall per capita CO₂ emissions, and the overall per capita GDP. 5. IQR, CV, and PSD refer to Inter-Quartile Range, Coefficient of Variation, and Pseudo Standard Deviation, respectively.

Table 3

Matrix of correlations.

Variables	(1)	(2)	(3)	(4)	(5)	(6)	(7)
(1) GVC	1.000						
(2) GVC_HH	0.838***	1.000					
(3) GVC_HL	0.600***	0.723***	1.000				
(4) GVC_LL	0.762***	0.837***	0.812***	1.000			
(5) GVC_LH	0.651***	0.734***	0.620***	0.690***	1.000		
(6) LNCO2PC	0.194***	0.207***	0.125***	0.174***	0.016	1.000	
(7) LNGDPPC	0.116***	0.208***	0.228***	0.202***	-0.145***	0.766***	1.000
Variables	(8)	(9)	(10)	(11)	(12)	(13)	
(8) GVC_H	1.000						
(9) LNCO2PC_H	-0.013	1.000					
(10) LNGDPPC_H	0.050	0.389***	1.000				
(11) GVC_L				1.000			
(12) LNCO2PC_L				-0.143**	1.000		
(13) LNGDPPC_L				0.184***	0.750***	1.000	

Notes: 1. GVC, GVC_HH, GVC_HL, GVC_LL, GVC_LH, GVC_L, and GVC_L stand for the overall global value chain participation degree and degrees in the HH, HL, LL, and LH industry groups as well as high (H) and low(L) income groups, respectively. 2. LNGDPPC, LNGDPPC_H, and LNGDPPC_L are the logarithm of GDP per capita in the full, high-income, and low-income sample, respectively. 3. LNCO2PC, LNCO2PC_H, and LNCO2PC_L stand for the logarithm of per capita CO₂ emission in the full, high-income, and low-income sample, respectively. 4. * p < 0.1, ** p < 0.05, *** p < 0.01.

Table 4

PVAR(1) model regression results.

	(1)	(2)	(3)
Vars.	GVC	LNGDPPC	LNCO2PC
L.GVC	0.410***	-0.007***	-0.012***
	(0.075)	(0.002)	(0.003)
L.LNGDPPC	-6.426***	0.959***	0.075
	(1.838)	(0.043)	(0.074)
L.LNCO2PC	3.042	-0.183^{***}	0.918***
	(2.111)	(0.041)	(0.072)

Obs. 567 (N = 63, T = 9).

Instruments: l(1/3).(GVC LNGDPPC LNCO2PC).

Hansen's J: 24.186 [0.149].

Notes: 1. Robust standard errors are in parentheses and *p*-values in brackets; 2. * p < 0.1, ** p < 0.05, *** p < 0.01.

The 1-period lag of LNCO2PC is positively correlated with GVC in both groups, while the coefficient in the high-income group is insignificant and smaller. (3) The 1-period lag of LNCO2PC is negatively correlated with LNGDPPC in both groups, while the coefficient is insignificant in the low-income group. The 1-period lag of LNGDPPC has an insignificant correlation with LNCO2PC in both groups.

4.3. IRF analysis

To further illustrate the dynamic relationships between the three variables and to examine causality, we simulated their OIRFs. The estimates in Tables 4–6 simply reflect reduced-form relationships between the variables. The OIRF analyzes the impact of a standard deviation shock¹¹ of the random disturbance term of a variable on the current and future states of other variables. The running sums of the OIRFs over the steps within each impulse–response pair are the cumulative OIRFs. In giving the variable a standard deviation shock, we used the Monte Carlo method to simulate the shock 200 times and obtained orthogonal

¹¹ A standard deviation shock is considered to be positive by default.

Table 5

PVAR(2) model regression results by industry group.

Panel I: HH	(1)	(2)	(3)		
Vars.	GVC_HH	LNGDPPC	LNCO2PC		
L.GVC_HH		-0.023***	-0.017*		
		(0.007)	(0.010)		
L2.GVC_HH		-0.004	0.004		
		(0.004)	(0.004)		
L.LNGDPPC	9.342*				
	(5.249)				
L2.LNGDPPC	-7.036**				
	(3.539)				
L.LNCO2PC	0.9				
	(2.893)				
L2.LNCO2PC	-3.409				
	(2.154)				
Instruments: 1(1/4).(GVC_HH LNGDPPC LNCO2PC)					
Hansen's J $chi2(18) = 2$	5.928323 [0.101]				

Panel II: HL	(1) CVC III	(2)	(3) LNCO2DC
L.GVC_HL	GVC_HL	-0.008	-0.019*
		(0.005)	(0.010)
L2.GVC_HL		0.001	-0.001
		(0.002)	(0.005)
L.LNGDPPC	7.128		
	(6.240)		
L2.LNGDPPC	-2.554		
	(3.850)		
L.LNCO2PC	5.214*		
	(2.766)		
L2.LNCO2PC	-7.999***		
	(2.320)		

Instruments: l(1/4).(GVC_HL LNGDPPC LNCO2PC) Hansen's J chi2(18) = 21.312039 [0.264]

Panel III: LL	(1)	(2)	(3)
Vars.	GVC_LL	LNGDPPC	LNCO2PC
L.GVC_LL		-0.008***	-0.003
		(0.002)	(0.005)
L2.GVC_LL		0.002	0.003
		(0.002)	(0.004)
L.LNGDPPC	3.821***		
	(1.400)		
L2.LNGDPPC	-3.764***		
	(1.286)		
L.LNCO2PC	-2.933***		
	(0.741)		
L2.LNCO2PC	-0.564		
	(0.579)		
Instruments: l(1/3).(GVC	LL LNGDPPC LNCO2PC)	
Hansen's J chi2(9) $= 17.5$	510961 [0.041]		

Panel IV: LH	(1)	(2)	(3)
Vars.	GVC LH	LNGDPPC	LNCO2PC
L.GVC_LH	-	-0.030***	-0.023
-		(0.010)	(0.015)
L2.GVC_LH		-0.002	0.003
		(0.005)	(0.007)
L.LNGDPPC	3.284		
	(2.896)		
L2.LNGDPPC	-0.56		
	(2.275)		
L.LNCO2PC	3.197**		
	(1.361)		
L2.LNCO2PC	-4.056***		
	(0.969)		
Instruments: l(1/4).(GVC	LH LNGDPPC LNCO2P	2)	
Hansen's J chi2(18) = 26	5.724276 [0.084]		

Notes: 1. Robust standard errors are in parentheses and p-values in brackets; 2. * p < 0.1, ** p < 0.05, *** p < 0.01; 3. Obs. 504 (N = 63, T = 8).

Table 6 PVAR(1) model regression results by income group

	(i) model regression results by meenie group;					
High income	(1)	(2)	(3)			
Vars.	GVC_H	LNGDPPC_H	LNCO2PC_H			
L.GVC_H	0.286***	-0.005*	-0.012^{***}			
	(0.110)	(0.003)	(0.004)			
L.LNGDPPC_H	8.276	0.712***	0.100			
	(5.789)	(0.172)	(0.204)			
L.LNCO2PC_H	3.955	-0.190***	0.860***			
	(2.763)	(0.048)	(0.076)			
Obs. $351(N = 39, T)$	= 9)					
Instruments: l(1/3	3).(GVC_H LNGDPPC_H	LNCO2PC_H)				
Hansen's J chi2(1	$(8) = 19.826745 \ (p = 0)$	0.343)				
Low income	(1)	(2)	(3)			
Vars	GVC L	LNGDPPC L	LNCO2PC L			
L GVC I	0.455***	_0.003	_0.002			
H.010_H	(0,100)	(0.002)	(0.002)			
L INCORPC I	10.627***	0.025	0.080			
L'ENGDLLC_F	(2.277)	(0.930	(0.072)			
L INCORD I	(2.2//)	(0.042)	(0.072)			
L.LINCOZPC_L	6.300**	-0.050	1.0/1			
	(3.205)	(0.063)	(0.101)			
Obs. 216 ($N = 24$, 1	Γ = 9)					
Instruments: l(1/3	3).(GVC_L LNGDPPC_L	LNCO2PC_L)				
Hansen's J chi2(1	(8) = 19.446107 (p = 0)	.365)				

Notes: 1. Robust standard errors are in parentheses and *p*-values in brackets; 2. * p < 0.1, ** p < 0.05, *** p < 0.01.

impulse response graphs and cumulative orthogonal impulse response graphs. 12

4.3.1. Baseline IRF results

Figs. 6.1 and 6.2 show the year-by-year and cumulative impulse response graphs, respectively. The first rows show the responses of LNCO2PC, LNGPDPC, and GVC to a standard deviation shock of LNCO2PC. We found that LNGDPPC has a significantly negative response to a standard deviation shock of LNCO2PC. The cumulative response of LNGDPPC to the shock in LNCO2PC is increasingly negative over the ten-year horizon.¹³ A standard deviation shock in LNCO2PC produces a sustained positive impact on GVC, and hence the cumulative response turns out to be increasingly positive.

The second rows show the responses of the three variables to a standard deviation shock of LNGDPPC. LNCO2PC positively responds to a standard deviation shock of LNGDPPC year-by-year, and the cumulative response is increasingly positive over the ten-year horizon. GVC has a negative year-by-year response to the shock on LNGDPPC and the response turns to become positive around the tenth year, thus showing a U-shape. The cumulative response is negative and converges to a negative value in the long run.

The third rows present the responses of the three variables to a standard deviation shock of GVC. A standard deviation shock of GVC increases LNCO2PC in the current period, but the year-by-year response of LNCO2PC then turns negative. The cumulative response of LNCO2PC to the shock in GVC is increasingly negative over the ten-year horizon. The year-by-year response of LNGDPPC to a standard deviation shock of GVC is positive at first, soon turns negative, and then changes to positive again in the fifth period. The cumulative response of LNGDPPC to the shock of GVC turns from negative to positive around the tenth year.

 $^{^{12}\,}$ The results obtained from a larger number of repetitions did not produce a significant difference.

¹³ Considering that the accuracy of the prediction decreases over time and that our sample covers only eleven years, it is reasonable that the impulse–response graphs in this study show the impulse–response dynamics for ten years (Figs 6.1 and 6.2) or eight years (Figs 7.1 and 7.2). The impulse–response graphs for longer periods are available upon request.



Fig. 6.1. Orthogonalized impulse response.



Fig. 6.2. Cumulative orthogonalized impulse response.



Fig. 7.1. Orthogonalized impulse response by industry group.

4.3.2. IRF results by industry group

Figs. 7.1 (a) to (d) and 7.2 (a) to (d) show the major year-by-year and cumulative impulse response graphs of industry groups, respectively, for the relationships of primary interest (see footnote 10). We found the following: (1) The year-by-year response of LNCO2PC to a standard deviation shock of GVC participation in the HL group is consistently negative, and thus, the cumulative response is increasingly negative during the ten years. The year-by-year response of LNGDPPC turns from negative to positive in the second year and the cumulative response is increasingly positive during the ten-year horizon. The year-by-year responses of LNCO2PC and LNGDPPC to a standard deviation shock of GVC participation in the HH and LH groups are continuously negative, and the cumulative responses are increasingly negative during the ten years. The cumulative response of LNGDPPC of the HH group is relatively strong, compared to the LH group. The year-by-year responses of LNCO2PC and LNGDPPC to a standard deviation shock of GVC participation in the LL group are positive at first, followed by a brief period of negative responses, and then fade away. The cumulative responses are positive and stable in the long run. (2) With a standard deviation shock of LNCO2PC, the year-by-year responses of GVC participation in the HL, HH, and LH groups are positive and increasing at first, and then gradually decreasing to become negative. The negative year-by-year response of the HL group is the largest in the long term, followed by the HH and LH groups. The cumulative responses of GVC participation

in these three groups are positive in the beginning and then become increasingly negative in the long term. The response of GVC participation in the LL group is generally negative and the impulse-response curve presents a U-shape. Consequently, the cumulative response is increasingly negative at first and converges to a negative value in the long run. (3) With a standard deviation shock of LNGDPPC, the year-byyear responses of GVC in the HH and LH groups are continuously positive and inversely U-shaped, and the cumulative responses are increasingly positive and then tend to become relatively stable in the long run. The year-by-year responses of GVC in the HL and LL groups are at first positive and increasing, and then decreasing to become negative. Differently, the year-by-year positive response of the LL group is relatively small and lasts for a relatively short time, compared to the HL group. Thus, the cumulative response of GVC in the HL group is persistently positive, while it is positive at first and becomes negative in the long term in the LL group.

4.3.3. IRF results by income group

Figs. 8.1(a) and (b) and 8.2 (a) and (b) show the year-by-year and cumulative impulse response graphs of income groups, respectively. We found the following: (1) With a standard deviation shock of GVC, the year-by-year and cumulative responses of LNCO2PC turn from positive to negative, while the year-by-year and cumulative responses of LNGDPPC turn from negative to positive in the high-income group. In



Fig. 7.2. Cumulative orthogonalized impulse response by industry group.



Fig. 8.1. Orthogonalized impulse response by income group.



(a) High-income Group

(b) Low-income Group

Fig. 8.2. Cumulative orthogonalized impulse response by income group.

the low-income group, the year-by-year and cumulative responses of LNCO2PC are continuously positive, while the year-by-year and cumulative responses of LNGDPPC turn from positive to negative. (2) In the high-income group, the year-by-year and cumulative responses of GVC to a standard deviation shock of LNCO2PC are initially increasingly positive and then drop to become negative. The year-by-year response of GVC to a standard deviation shock of LNGDPPC is increasingly positive at first and then decreases to zero, and the cumulative response is increasingly positive. In the low-income group, the year-by-year and cumulative responses of GVC to a standard deviation shock of LNCO2PC are increasingly positive, while the responses to a standard deviation shock of LNGDPPC are increasingly negative. (3) In the high-income group, the year-by-year and cumulative responses of LNCO2PC to a standard deviation shock of LNGDPPC are inverted U-shaped and are negative in the long term. The year-by-year response of LNGDPPC to a standard deviation shock of LNCO2PC is increasingly negative at first and the negative value drops to zero in the long run; thus, the cumulative response is increasingly negative. In the low-income group, the year-by-year and cumulative responses of LNCO2PC to a standard deviation shock of LNGDPPC are increasingly positive, while the reverse responses are increasingly negative.

4.4. FEVD analysis

To compare the effect sizes, which reflect the relative importance of different shocks on each of the three variables, we performed a variance decomposition of the PVAR model. We repeated the Monte Carlo simulation 200 times to simulate the first eight forecast periods of the variance decomposition. Tables 7–9 show the results of periods one, four, and eight.

Table 7

Forecast-error va	riance de	ecomposition	n.
-------------------	-----------	--------------	----

Response var.		Impulse var.			
	Period	GVC	LNGDPPC	LNCO2PC	
	1	100%	0%	0%	
GVC	4	88.277%	3.821%	7.901%	
	8	65.198%	5.345%	29.458%	
	1	3.707%	96.293%	0%	
LNGDPPC	4	6.281%	60.614%	33.105%	
	8	2.345%	28.138%	69.518%	
	1	2.667%	0.196%	97.136%	
LNCO2PC	4	8.405%	2.216%	89.379%	
	8	15.947%	12.048%	72.005%	

Forecast-error variance decomposition by industry group.

Panel I: HH		Impulse var.		
Response var.	Period	GVC_HH	LNGDPPC	LNCO2PC
	1		0%	0%
GVC_HH	4		24.036%	16.306%
	8		17.714%	38.613%
	1	4.140%		
LNGDPPC	4	15.093%		
	8	19.459%		
	1	2.781%		
LNCO2PC	4	12.938%		
	8	30.154%		
Panel II: HL		Impulse var.		
Response Var.	Period	GVC_HL	LNGDPPC	LNCO2PC
	1		0%	0%
GVC_HL	4		10.860%	15.471%
	8		6.801%	46.912%
	1	2.688%		
LNGDPPC	4	0.581%		
	8	8.211%		
	1	0.001%		
LNCO2PC	4	16.618%		
	8	23.064%		
Panel III: LL		Impulse var.		
Response var.	Period	GVC_LL	LNGDPPC	LNCO2PC
				00/
	1		0%	0%
GVC_LL	1 4		0% 0.894%	0% 4.734%
GVC_LL	1 4 8		0% 0.894% 0.967%	0% 4.734% 4.770%
GVC_LL	1 4 8 1	6.840%	0% 0.894% 0.967%	0% 4.734% 4.770%
GVC_LL LNGDPPC	1 4 8 1 4	6.840% 4.278%	0% 0.894% 0.967%	0% 4.734% 4.770%
GVC_LL LNGDPPC	1 4 8 1 4 8	6.840% 4.278% 4.273%	0% 0.894% 0.967%	0% 4.734% 4.770%
GVC_LL LNGDPPC	1 4 8 1 4 8 1	6.840% 4.278% 4.273% 0.714%	0% 0.894% 0.967%	0% 4.734% 4.770%
GVC_LL LNGDPPC LNCO2PC	1 4 8 1 4 8 1 4	6.840% 4.278% 4.273% 0.714% 1.204%	0% 0.894% 0.967%	0% 4.734% 4.770%
GVC_LL LNGDPPC LNCO2PC	1 4 8 1 4 8 1 4 8	6.840% 4.278% 4.273% 0.714% 1.204% 1.202%	0% 0.894% 0.967%	0% 4.734% 4.770%
GVC_LL LNGDPPC LNCO2PC	1 4 8 1 4 8 1 4 8 1 4 8	6.840% 4.278% 4.273% 0.714% 1.204% 1.202%	0% 0.894% 0.967%	0% 4.734% 4.770%
GVC_LL LNGDPPC LNCO2PC Panel IV: LH	1 4 8 1 4 8 1 4 8	6.840% 4.278% 4.273% 0.714% 1.204% 1.202% Impulse var.	0% 0.894% 0.967%	0% 4.734% 4.770%
GVC_LL LNGDPPC LNCO2PC Panel IV: LH Response var.	1 4 8 1 4 8 1 4 8 7 Period	6.840% 4.278% 4.273% 0.714% 1.204% 1.202% Impulse var. GVC_LH	0% 0.894% 0.967% <i>LNGDPPC</i>	0% 4.734% 4.770% <i>LNCO2PC</i>
GVC_LL LNGDPPC LNCO2PC Panel IV: LH Response var.	1 4 8 1 4 8 1 4 8 <i>Period</i> 1	6.840% 4.278% 4.273% 0.714% 1.204% 1.202% Impulse var. GVC_LH	0% 0.894% 0.967% <i>LNGDPPC</i> 0%	0% 4.734% 4.770% <i>LNCO2PC</i> 0%
GVC_LL LNGDPPC LNCO2PC Panel IV: LH Response var. GVC_LH	1 4 8 1 4 8 1 4 8 7 Period 1 4	6.840% 4.273% 4.273% 0.714% 1.204% 1.202% Impulse var. GVC_LH	0% 0.894% 0.967% <i>LNGDPPC</i> 0% 22.058%	0% 4.734% 4.770% <i>LNCO2PC</i> 0% 6.640%
GVC_LL LNGDPPC LNCO2PC Panel IV: LH Response var. GVC_LH	1 4 8 1 4 8 1 4 8 Period 1 4 8	6.840% 4.278% 4.273% 0.714% 1.204% 1.202% Impulse var. GVC_LH	0% 0.894% 0.967% <i>LNGDPPC</i> 0% 22.058% 18.813%	0% 4.734% 4.770% <i>LNCO2PC</i> 0% 6.640% 34.569%
GVC_LL LNGDPPC LNCO2PC Panel IV: LH Response var. GVC_LH	1 4 8 1 4 8 1 4 8 7 <i>Period</i> 1 4 8 1	6.840% 4.278% 4.273% 0.714% 1.204% 1.202% Impulse var. GVC_LH 2.844%	0% 0.894% 0.967% <i>LNGDPPC</i> 0% 22.058% 18.813%	0% 4.734% 4.770% 4.770% <i>LNCO2PC</i> 0% 6.640% 34.569%
GVC_LL LNGDPPC LNCO2PC Panel IV: LH Response var. GVC_LH LNGDPPC	1 4 8 1 4 8 1 4 8 7 <i>Period</i> 1 4 8 1 4	6.840% 4.278% 4.273% 0.714% 1.204% 1.202% Impulse var. GVC_LH 2.844% 38.882%	0% 0.894% 0.967% <i>LNGDPPC</i> 0% 22.058% 18.813%	0% 4.734% 4.770% 4.770% <i>LNCO2PC</i> 0% 6.640% 34.569%
GVC_LL LNGDPPC LNCO2PC Panel IV: LH Response var. GVC_LH LNGDPPC	1 4 8 1 4 8 1 4 8 7 8 7 8 7 8 1 4 8	6.840% 4.278% 4.273% 0.714% 1.204% 1.202% Impulse var. GVC_LH 2.844% 38.882% 48.314%	0% 0.894% 0.967% <i>LNGDPPC</i> 0% 22.058% 18.813%	0% 4.734% 4.770% 4.770% <i>LNCO2PC</i> 0% 6.640% 34.569%
GVC_LL LNGDPPC LNCO2PC Panel IV: LH Response var. GVC_LH LNGDPPC	1 4 8 1 4 8 1 4 8 <i>Period</i> 1 4 8 1 4 8 1	6.840% 4.278% 4.273% 0.714% 1.204% 1.202% Impulse var. GVC_LH 2.844% 38.882% 48.314% 0.012%	0% 0.894% 0.967% <i>LNGDPPC</i> 0% 22.058% 18.813%	0% 4.734% 4.770% <i>LNCO2PC</i> 0% 6.640% 34.569%
GVC_LL LNGDPPC LNCO2PC Panel IV: LH Response var. GVC_LH LNGDPPC LNCO2PC	1 4 8 1 4 8 1 4 8 7 Period 1 4 8 1 4 8 1 4	6.840% 4.273% 0.714% 1.204% 1.202% Impulse var. GVC_LH 2.844% 38.882% 48.314% 0.012% 27.978%	0% 0.894% 0.967% <i>LNGDPPC</i> 0% 22.058% 18.813%	0% 4.734% 4.770% 4.770% <i>LNCO2PC</i> 0% 6.640% 34.569%

Table 9

Forecast-error variance decomposition by income group.

High-income		Impulse var.		
Response var.	Period	GVC_H	LNGDPPC_H	LNCO2PC_H
	1	100.000%	0.000%	0.000%
GVC_H	4	93.310%	4.447%	2.243%
	8	91.944%	4.872%	3.183%
	1	0.103%	99.897%	0.000%
LNGDPPC_H	4	6.956%	58.726%	34.318%
	8	6.366%	39.580%	54.054%
	1	4.224%	0.007%	95.768%
LNCO2PC_H	4	9.147%	0.178%	90.675%
	8	10.744%	0.343%	88.913%
Low-income		Impulse var.		
Response var.	Period	GVC_L	LNGDPPC_L	LNCO2PC_L
	1	100.000%	0.000%	0.000%
GVC_L	4	77.723%	3.598%	18.680%
	8	39.708%	5.274%	55.018%
	1	12.070%	87.930%	0.000%
LNGDPPC_L	4	4.043%	86.172%	9.784%
	8	2.332%	50.413%	47.255%
	1	4.121%	1.052%	94.827%
LNCO2PC_L	4	1.617%	0.528%	97.855%
	8	0.958%	0.259%	98.783%

4.4.1. Baseline FEVD results

Observing Table 7, we found that: (1) The variation of each of the three variables is largely due to their disturbances, and the importance of these disturbances decreases over time. (2) Eight periods ahead, the variations of GVC and LNGDPPC explain approximately 16% and 12%, respectively, of the variation in LNCO2PC, increasing at longer periods. LNCO2PC in turn contributes to the GVC and LNGDPPC, again increasing at longer periods. After eight periods, LNCO2PC explains around 29.5% and 69.5% of the total variations in GVC and LNGDPPC, respectively. (3) LNGDPPC explains only about 5.3% of the variation in GVC eight periods ahead, and the effect is stable over time. The contribution of GVC to the variation in LNGDPPC is as low as 3.7% one period ahead, and the magnitude grows to nearly 6.3% four periods ahead and then decreases to 2.3% eight periods ahead, thereby displaying an inverted U-shape.

4.4.2. FEVD results by industry group

Table 8 presents the major variance decomposition results by industry group for the relationships of primary interest (see footnote 10). We mainly found the following: (1) Eight periods ahead, GVC in HH and LH explains 19.4% and 48.3% of the variation of LNGDPPC, greater than the variation explained for short periods, while GVC in HL and LL explains only 8.2% and 4.3% of the variation in LNGDPPC, respectively. (2) The contributions of GVC in HH and HL to the variation in LNCO2PC eight periods ahead are 30.2% and 23.1%, respectively, and are increasing over time, while those of GVC in LL and LH are 1.2% and 26.3%, respectively, in a decreasing manner. (3) Observing the eighth period, we found that LNCO2PC accounts for 46.9% and 38.6% of the variation in GVC in HL and HH, respectively, followed by LH (34.6%) and LL (4.8%). LNGDPPC explains 18.8% and 17.7% of the variation in GVC in LH and HH, respectively, while it explains only 6.8% and 0.97% of the variation in GVC in HL and LL, respectively.

4.4.3. FEVD results by income group

Table 9 presents the variance decomposition results by income group. We mainly found the following: (1) GVC explains 10.7% of the variation in LNCO2PC eight periods ahead in the high-income group in an increasing manner, while it explains only 0.96% of the variation inLNCO2PC in the eighth period in the low-income group in a decreasing manner. Moreover, the contribution of GVC to the variation in LNGDPPC eight periods ahead in the high-income group is 6.4% and is generally increasing over time, while in the low-income group, the contribution is 2.3% in a decreasing manner. (2) Observing the eighth period, we found that LNCO2PC accounts for 3.2% and 55.02% of the GVC variation in the high- and low-income groups, respectively. LNGDPPC explains 4.9% and 5.3% of the GVC variation in the high- and low-income groups, respectively. (3) Observing the eighth period, we found that the contributions of LNGDPPC to the variations in LNCO2PC in the high- and low-income groups are 0.343% and 0.259%, respectively. Moreover, LNCO2PC accounts for 54.054% and 47.255% of the variations in LNGDPPC in the high- and low-income groups.

4.5. Robustness checks

We performed three robustness checks. First, we checked the sensitivity of our impulse responses and variance decompositions to the ordering of the three variables. Second, we assessed the potential of spatial correlation by spatial clustering of the errors. Third, we examined the sensitivity of our results to specifying a lag length of one period rather than two. These results for robustness checks are available on request.

4.5.1. Reorder of variables in the model

The impulse response results of the panel VAR could be affected by the order of variables in the model. To check the robustness of our results with the order of {GVC, LNGDPPC, LNCO2PC}, we tried the other five possible orders and found the following: (1) At the aggregate level, the results of the PVAR regressions, year-by-year impulse-response (OIRF) dynamics, cumulative impulse-response (COIRF) dynamics, and forecast-error variance decompositions (FEVDs) are generally in line with those shown in this paper. (2) As GVC participation in different industry groups is considered, the PVAR regression results are consistent. The OIRF and COIRF dynamic results are robust for all groups except for the LL group. As the order changes, the results of the LL group are slightly different from those we presented in this paper and the differences occur in the first or second periods. Further, the results of FEVD are mostly robust, and slight differences are mainly found in the HL and LL groups. (3) As income groups are considered, the PVAR regression results are consistent. The OIRF and COIRF dynamic results are robust for all groups, especially for the high-income group. The results of FEVD are also mostly robust, and slight differences are mainly found in the low-income group.

4.5.2. Spatial dependence

Given some evidence of cross-sectional dependence, one solution is to apply a recently developed spatial panel VAR modeling approach introduced by Civelli et al. (2018) for a two-variable case. However, as noted by Civelli et al. (2018, p. 81), "...the spatial terms are not exogenous by construction because they are correlated with the current innovations to the endogenous variables." The issue of causality is problematic in spatial lag models and not readily solvable with instrumental variables (Gibbons and Overman, 2012).¹⁴ Because our focus is not estimating the spatial spillovers among countries, and the ordering of causality among countries in impulse response analysis and variance decompositions would be somewhat arbitrary, adding significant complexity to the ordering of the innovations in the three variables, we do not estimate a spatial panel model.

But to mitigate the effects of spatial correlation that may affect the baseline results, we attempted to account for cross-sectional dependence by using a clustered standard error approach that grouped 63 countries and regions by their continents. The approach mitigates the potential effects of spatial correlation on estimated standard errors and increases the efficiency of the slope estimates in the baseline results. We found the

¹⁴ Time-lagged spatial variables may not better capture the relationships between spatial units than current-period variables. The time-lagged variables may simply be more exogenous econometrically but not reflect true causality.

following: (1) At the aggregate level, though the scale of some coefficients changed, the results of the PVAR regressions and FEVDs are generally robust. The changes in ORIF mainly take place in the medium run and thus the convexity of ORIF only slightly differs, while the CORIF is relatively more robust. (2) As the industry groups are considered, besides the changes in the scale of coefficients of PVAR(2) regressions, a few coefficients' significance and/or signs have changed for the HH, HL, and LH groups. The ORIFs, CORIFs, and FEVDs of these groups are slightly different as well. (3) When considering income groups, the scale of coefficients from PVAR(1) regressions changed. Except for the effect of the 1-period lag of LNGDPPC on GVC that turned from insignificantly positive to significantly negative in the high-income group, the signs of other coefficients are generally robust. The ORIFs and CORIFs are consistent in the low-income group, while slightly different in the highincome group, especially in the short- and medium-run. The FEVD results are slightly different as well, especially in the high-income group. The results showed that high-income countries are more likely to exhibit spatial correlation.

4.5.3. Specifying a different lag length

Referring to GVC participation in different industry groups, we presented results based on 2-period lags. While the results of MAIC, MBIC, and MQIC showed that the 1-period lag regressions are preferred in the HL and LL groups, the results of MBIC and MQIC showed that the 1-period lag is preferred in the HH group and the result of MQIC revealed that the 1-period lag is preferred in the LH group. Therefore, we tried the PVAR(1) regressions as well as ORIF, CORIF, and FEVD in industry group analyses and did not find significant differences.

In sum, our results are generally robust and the aforementioned hypotheses (H1, H2, H3a, and H3b) are verified by our results.

5. Discussion

5.1. The effects of GVC on LNCO2PC and LNGDPPC

The empirical results show that: (1) GVC will reduce LNCO2PC, consistent with Yao et al. (2021), and increase LNGDPPC in the long term, in line with Kowalski et al. (2015) and Ignatenko et al. (2019), though it may decrease LNGDPPC in the short term. They are the combined results of the scale, technique spillover, composition, and competition effects, as discussed in Wang et al. (2019). Some of the effects are positively associated with LNCO2PC and LNGDPPC, while others are negatively associated. In the long run, GVC's negative impact on LNCO2PC and the positive impact on LNGDPPC dominate. Additionally, the deeper the GVC participation degree, the more a participant country will be affected by international markets. When a GVCparticipating country suffers adverse shocks, the negative impacts will spread to other GVC participants, and vice versa. During the period of interest, the economies of most countries in the sample were severely affected by the 2008 global financial crisis, and many of them took a long time to recover. To a certain degree, this probably explains the short-run adverse effect of GVC on LNGDPPC.

(2) GVC participation in different industries will affect LNCO2PC heterogeneously, similar to the results of Wang et al. (2019). GVC participation in high-CO₂ industries (HH and HL) shows higher impacts on LNCO2PC than in low-CO₂ industries (LH and LL). GVC participation in different industries will also affect LNGDPPC heterogeneously, consistent with Kordalska et al. (2016) and Formai and Vergara Caffarelli (2015). GVC participation in high value-added industries shows higher impacts on LNGDPPC. This result was expected because high-CO₂ industries are relatively sensitive to CO₂-related shocks, and low value-added industries have smaller impacts on a country's GDP, and vice versa.

Specifically, in the long run, GVC participation in high-CO₂ industries (HH and HL) will reduce LNCO2PC, and that in low value-added industries (HL and LL) will enhance LNGDPPC. GVC participation in HL industries may be the best scenario because it will reduce LNCO2PC while enhancing LNGDPPC. Moreover, GVC participation in LH industries can reduce LNCO2PC but worsen LNGDPPC. This result proves that the scale, technique spillover, composition, and competition effects play heterogeneous roles in different industries, which is reasonable considering the various production characteristics of industries.

(3) The responses of LNCO2PC and LNGDPPC to GVC vary by income. The responses in the high-income group are stronger than those in the low-income group, which is in line with the findings of Kummritz (2015) and Fagerberg et al. (2018). Further, in the high-income group, the response of LNCO2PC is generally negative and that of LNGDPPC turns from negative to positive. In the low-income group, the response of LNCO2PC is generally positive and that of LNGDPPC turns from positive to negative. In other words, in the long run, through participating in GVCs, high-income countries could benefit in terms of higher LNGDPPC and lower LNCO2PC, while both economic and carbon emission conditions worsen in low-income countries.

High-income countries usually have comparative advantages in high-tech and capital-intensive commodities. With participation in GVCs, high-income countries can focus on producing these commodities by transferring labor-intensive and energy-intensive production abroad, as discussed by Duan et al. (2021). The composition and scale effects lead to a reduction in LNCO2PC but may result in the increase or decrease of LNGDPPC. Further, according to Ivarsson and Alvstam (2010), Zhang and Gallagher (2016), and Reddy et al. (2020), GVCs make the spread of technologies worldwide easier and faster; to keep comparative advantages in intensive international competition, highincome countries are motivated to continuously innovate and upgrade technologies, which is conducive to their economies and natural environments. In brief, the responses of LNCO2PC and LNGDPPC to GVC are the comprehensive results of scale, composition, technique spillover, and competition effects.

Low-income countries generally have comparative advantages in labor- and energy-intensive commodities. By participating in GVCs, the scale and composition effects expand the production of these commodities, which stimulates economic growth while exacerbating carbon emissions. The positive effects of technology spillovers on economic growth and carbon reduction might not occur in low-income countries. First, measures, such as patent protection, have prevented high-income countries from spilling over technologies to low-income countries (Sanyal, 2004; Saggi, 2007). Second, the R&D, education, and infrastructure of low-income countries are usually in poor condition, which may not support the effective application of advanced technologies. Moreover, the competition effect may force low-income countries to reduce their high-polluting production to meet international environmental standards, but they may not be capable of expanding environmentally-friendly production. Therefore, low-income countries may not be able to directly benefit from participating in GVCs for a considerable period.

5.2. The effects of LNCO2PC and LNGDPPC on GVC

(1) LNCO2PC will positively affect GVC, which can be due to the carbon transfer effect (Zhang et al., 2017; Ben-David et al., 2018) and environmental protection effect (Do Paço et al., 2009; Glucker et al., 2013). On the one hand, countries may use GVC participation as an effective measure to deal with CO_2 emissions (Wang et al., 2019). On the other hand, the increased production, or the lower environmental standards, which result in higher LNCO2PC, may be conducive to GVC participation, especially in low-income countries. LNGDPPC will negatively affect GVC. As discussed, GVC participation may stimulate or reduce GDP growth. If the stimulation effect dominates, countries will take measures to enhance GVC participation, otherwise, they will reduce GVC participation. A possible explanation for the negative effect of LNGDPPC on GVC is that the unfavorable aspects of GVC participation are gradually emerging as the anti-globalization sentiment increases

(Meyer, 2017; Branicki et al., 2021).

(2) GVC participation in different industries shows heterogeneous responses to changes in LNCO2PC and LNGDPPC. As LNCO2PC increases, GVC participation in high-CO2 industries (HH and HL) as well as low-CO2, high value-added industries (LH) will increase at first and then decrease in the long run. The positive response may result from the negative effect of GVC participation in these industries on LNCO2PC. The consequent negative response implies, to a certain extent, that GVC participation in these industries may be an effective but not optimal way to reduce LNCO2PC in the long run. A possible reason is an adverse effect of participating in GVCs. Increased participation in GVCs raises the dependence on international markets and the susceptibility to global economic fluctuations. Therefore, in the long run, countries tend to find better CO₂ solutions to replace GVC participation, such as clean energy and clean technologies. Moreover, GVC participation in low-CO₂, low value-added (LL) industries will decrease as LNCO2PC increases, because it will aggravate CO2 emissions, as discussed above.

LNGDPPC positively affects GVC in all the groups during a relatively short period, and thereafter, negatively affects them over a relatively long period, thus showing an inverted U-shape. Further, the positive responses of GVC in the high value-added groups (HH and LH) are relatively larger and last relatively longer. The responses of GVC in the low value-add groups (HL and LL) become negative relatively earlier and the negative scales are relatively larger. Considering that the cumulative effects of GVC in high value-added groups on LNGDPPC are negative while the cumulative responses of GVC in those groups to LNGDPPC are positive, we believe that high value-added industries (HH and LH) mainly receive non-economic benefits (e.g., carbon reductions) from GVC participation, which is in line with the negative effects of GVC participation in HH and LH on LNCO2PC discussed above.

The cumulative response of GVC participation in HL industries to LNGDPPC is positive. This result is in line with our expectations since GVC participation in HL industries can promote economic growth and meanwhile reduce CO_2 emissions as discussed above. The cumulative effect of GVC in LL on LNGDPPC is positive while the cumulative response of GVC in LL industries to LNGDPPC turn from positive to negative. It implies that although GVC participation promotes the economic growth of LL industries, it brings about adverse effects on non-economic fields, such as CO_2 emissions, which is in line with the positive cumulative effect of GVC participation in LL on LNCO2PC discussed above. As economies develop, countries will pay more attention to the environment.

(3) The responses of GVC to the changes of LNCO2PC and LNGDPPC are different across income groups. The impact of LNCO2PC on GVC turns from positive to negative in the high-income group. This result again confirms our argument that GVC participation can be effective but might not be optimal in reducing CO_2 emissions. The impact of LNGDPPC on GVC is generally positive. As economies develop, high-income countries will benefit more from GVCs because they will be more powerful to take advantage of the global market. As the positive effects of GVC participation overcome the adverse effects, high-income countries will enhance GVC participation.

The response of GVC to LNCO2PC is generally positive in low-income countries, although GVC participation will enhance their LNCO2PC. This situation is largely because GVC participation can bring about economic benefits to low-income countries (Gereffi, 1999; Humphrey and Schmitz, 2002), which explains the short-term increase of GVC with LNGDPPC to a certain degree. Low-income countries tend to give priority to economic growth (Liu et al., 2018) even at the expense of the environment. However, in the long run, when the negative effects of GVC on LNGDPPC play leading roles in low-income countries (Fagerberg et al., 2018), GVC participation declines as LNGDPPC increases. This result shows that as their economies develop to a certain extent, low-income countries will reduce GVC participation to avoid the negative impacts on their economies and environments. Higher LNGDPPC will provide low-income countries with more opportunities to introduce

better measures, such as advanced technologies and clean energy, to protect the environment while developing their economies. Moreover, low-income countries may lose their low-cost comparative advantages from economic growth but have not yet developed new comparative advantages. In this case, it seems that their GVC participation degrees will be lower as economies develop since their GVC positions may be replaced by other low-income countries (Song and Wang, 2017).

5.3. The dynamics between LNCO2PC and LNGDPPC

Similar to those who have confirmed the existence of an EKC (Grossman and Krueger, 1991; Ang, 2007; Charfeddine, 2017; You and Lv, 2018), the response of LNCO2PC to LNGDPPC is found to be generally positive over time and presents an inverted U-shaped relationship. It indicates that in the early stage of economic growth, the scale effect dominates the effect of LNGDPPC on LNCO2PC. In this stage, the energy consumption required by economic development will stimulate LNCO2PC. As time goes by, economic development will promote technological progress, industrial composition adjustment, and environmental protection awareness, thus inhibiting CO₂ emissions. We may reasonably speculate that economic growth will result in CO₂ reduction if observed over a longer period. Referring to the effect of LNCO2PC on LNGDPPC, we found that it is generally negative, in line with the findings of Magazzino (2016a) on the South Caucasus area and those of Magazzino (2016b) on GCC countries, and presents a U-shaped curve. This result implies that reducing production may lower CO₂ emissions in the short run, but it is not an optimal carbon reduction measure and will not be chosen if better feasible measures are available to fix carbonrelated issues.

When different income groups are considered, the dynamic relationships vary substantially. In the high-income group, the response of LNCO2PC to LNGDPPC presents an inverted U-shape, which is consistent with the baseline result. It also turns from positive to negative over time, thereby confirming our speculation discussed above. In the lowincome group, the response of LNCO2PC to LNGDPPC is increasingly positive, thus showing that the technique spillover and competitive effects may not be fully utilized by low-income countries. These countries tend to prioritize economic growth; therefore, the negative impacts of the scale and composition effects on the environment dominate. The effects of LNCO2PC on LNGDPPC are generally negative in both groups, however, the effect presents a U-shape in the high-income group, while it is decreasingly negative in the low-income group. We have discussed that reducing production can reduce emissions, but it is not the best way. Clean technology and clean energy are better measures to reduce emissions (Soytas et al., 2007; Sbia et al., 2014; Magazzino, 2016d). However, their applications require that economies advance to a certain stage of development. Low-income countries may have not crossed those thresholds and must sacrifice production if they choose to reduce carbon emissions.

6. Conclusions and policy implications

This study applied the PVAR approach, using data for 63 countries and regions over the period 2005–2015, to analyze the dynamic relationships between GVC participation, CO_2 emissions, and economic growth and the underlying mechanisms of these relationships. Special attention was paid to the impacts of GVC participation on CO_2 emissions and GDP, in an attempt to unveil the relationship between GVC participation and sustainable growth. The empirical results support all hypotheses proposed. The major findings are: First, GVC participation is conducive to environmentally-friendly growth. The year-by-year and cumulative responses of per capita GDP to a standard deviation shock of GVC are positive in the long run, while those of per capita CO_2 emissions are negative. The FEVD results show that GVC participation explains 6.3% of the per capita GDP variation four periods ahead, and 16% of the variation of per capita CO_2 emissions eight periods ahead. These results are attributed to the mixed results of scale, technique spillover, composition, and competition effects. Second, we found evidence of heterogeneous effects between different industry groups. GVC participation in high-CO₂ emission industries has greater impacts on CO₂ emissions, and that in high value-added industries have greater impacts on GDP. GVC participation in high CO2 emission and low value-added industries contributes to economic growth and carbon reductions. GVC participation in other industries contributes either to economic growth or carbon reductions. Heterogeneous characteristics of industries make the aforementioned four effects work differently. Third, high-income countries benefit more from GVC participation in terms of economic growth and CO2 reductions. Several factors prevent lowincome countries from thoroughly benefiting from GVC participation, such as comparative disadvantages in capital-intensive and high-tech production, inefficient application of advanced technologies, and weak reactions to adverse shocks from international markets.

Other findings include: First, GVC participation has a long-run positive response to per capita CO₂ emissions and a negative response to per capita GDP. The variation of per capita CO₂ emissions explains a larger share of GVC variations, especially in high-CO2 groups. These results may imply that participating in GVCs is a feasible measure to deal with emissions, but that countries would reduce GVC participation as they become increasingly developed, probably due to the side effects brought by high dependencies on international markets. Second, the impact of per capita CO2 emissions on GVC participation varies by industry and income, so does per capita GDP. Specifically, per capita CO₂ emissions present a larger impact on GVC participation in high-CO2 industries, and per capita GDP presents a larger impact on GVC participation in high value-added industries, which is to be expected considering the industries' characteristics. Moreover, both per capita CO2 emissions and per capita GDP have larger impacts on the GVC participation of lowincome countries. A possible reason is that these countries have relatively lower GVC participation degrees, and hence are easier to have larger changes. Another possible reason is that low-income countries may not have better alternatives to deal with economic or environmental shocks. Third, per capita CO₂ emissions increase as per capita GDP increases, and an increase in per capita CO₂ emissions will lead to a decrease in per capita GDP. This dynamic relationship varies by income. The impact of per capita GDP on per capita CO₂ emissions presents an inverted U-shape and is negative in the long term in high-income countries, while it is increasingly positive in low-income countries. The impact of per capita CO₂ emissions on per capita GDP is negative, and presents a U-shape in high-income countries, while it is increasingly negative in low-income countries. The impact of per capita GDP on per capita CO₂ emissions, in line with existing studies, is the comprehensive result of four effects. Countries would rather reduce production to deal with CO₂ emissions, which reflects the growing public awareness of the environment and the lack of better alternative measures to tackle emissions, especially in low-income countries.

Accordingly, important policy implications emerge as follows: First, countries should take measures to actively participate in GVCs, such as lowering tariffs and non-tariff barriers, encouraging international communication, enhancing mutual understanding, etc. Meanwhile, countries should pay attention to the knock-on effects of trading partners suffering negative shocks from events such as financial crises, wars, or a pandemic. Lowering or withdrawing from GVCs is not the optimal solution to deal with the adverse effects. While integrating into international markets, countries must not neglect the development of their domestic markets. It is critical to maintain some independence in the domestic economy and not rely too heavily on trading partners. China's domestic and international dual-cycle development strategy could be a case for reference. Second, during the process of participating in GVCs, high-income countries should continue to strengthen their comparative advantages in technology and capital-intensive production and lower technological thresholds, promoting technology spillovers to lowincome countries. Low-income countries should improve domestic

human capital and infrastructure to efficiently utilize the technique spillover effects. In addition, low-income countries should optimize domestic production composition, shifting their competitiveness from low environmental standards to high-quality clean products. Third, countries should not blindly favor GVC participation in industries with high value added and low carbon emissions. In cases where GVC participation lowers CO_2 emissions, the effect of carbon reduction is more pronounced when countries participate in the GVCs of high-carbon emission industries. Similarly, countries should be aware that the domestic economy is more likely to be affected by adverse shocks on the international market when they participate in the GVC of high value added industries.

Finally, reducing production is not the optimal solution to tackle environmental problems. Instead, improving energy efficiency and promoting energy diversity are better solutions (Omri, 2013; Magazzino, 2016d)). Specifically, countries should increase investments in R&D and education, encouraging independent innovation in energy-saving technologies (Magazzino and Cerulli, 2019); actively use clean and renewable energy, such as solar, wind, and nuclear energy; and enhance public awareness of low-carbon production and consumption. Governments should set clear carbon reduction targets (Magazzino, 2015). For example, China proposed that CO₂ emissions will peak around 2030, the proportion of non-fossil energy in primary energy consumption will increase to about 20% by 2030, and CO2 emissions per GDP will drop by 60-65% compared with 2005 in 2030. Moreover, policymakers can supervise environmental pollution behaviors through legislation and regulations, such as carbon tax. They can also allocate special funds to financially motivate low-carbon behaviors, like subsidies for purchasing electric vehicles.

7. Limitations and future research

This study is not without limitations. First, although the results of the unit root tests in this study generally support (trend) stationarity, there were a few exceptions when allowing for cross-sectional dependence. Tobler's first law of geography pointed out that "everything is related to everything else, but near things are more related than distant things" (Tobler, 1970). To check the robustness of our results, we considered spatial interdependence by spatially clustering the errors, assuming that unobserved components in outcomes for units (countries) within clusters (continents) are correlated, but we did not estimate spatial spillovers. Second, countries' positions on the GVCs are different, which implies they are responsible for different phases of production and thus in turn lead to different CO_2 emissions and economic growth. This is an issue unexplored in this paper. Third, this study is limited by data availability.

Future research could attempt to account for cross-sectional dependence where it occurs by constructing a spatially correlated matrix (Zhu et al., 2022) and estimating spatial spillovers (e.g., the spatial panel VAR model as used in Civelli et al. (2018)), though the issue of identifying causal relationships across space remains a challenge. Further improvements could include incorporating the position index of a country's industries within GVCs (Koopman et al., 2014; Liu et al., 2018; Ye et al., 2020), and covering more countries and regions over a longer period.

Acknowledgements

This study was supported by Chongqing Social Science Planning Project (CN) (Project No. 2020PY39) and the National Natural Science Fund of China (Project No. 72073137). We appreciate the valuable comments from the editor and anonymous referees that have helped to significantly improve the paper. Nevertheless, the views expressed herein are those of the authors and we are, of course, responsible for all errors.

J. Wang et al.

References

- Abrigo, M., Love, I., 2016. Estimation of panel vector autoregression in Stata. Stata J. 16 (3), 778–804.
- Aklin, M., 2016. Re-exploring the trade and environment Nexus through the diffusion of pollution. Environ. Resour. Econ. 64 (4), 663–682.
- Ang, J., 2007. CO₂ emissions, energy consumption, and output in France. Energy Policy 35 (10), 4772–4778.
- Antonakakis, N., Cunado, J., Filis, G., Perez De Gracia, F., 2017. Oil dependence, quality of political institutions and economic growth: a panel VAR approach. Resour. Policy 53, 147–163.
- Antweiler, W., Copeland, B., Taylor, M., 2001. Is free trade good for the environment? Am. Econ. Rev. 91 (4), 877–908.
- Arce González, G., Cadarso Vecina, M., López Santiago, L., Tobarra Gómez, M., Zafrilla-Rodríguez, J., 2012. Indirect pollution Haven hypothesis in a context of global value chain. In: Final WIOD Conference: Causes and Consequences of Globalization, Groningen, The Netherlands.
- Ben-David, I., Kleimeier, S., Viehs, M., 2018. Exporting pollution: where do multinational firms Emit CO₂?. In: NBER Working Papers 25063. NBER, Inc.
- Borck, J.C., Coglianese, C., 2009. Voluntary environmental programs: assessing their effectiveness. Annu. Rev. Environ. Resour. 34 (1), 305–324.
- Branicki, L., Sullivan-Taylor, B., Brammer, S., 2021. Towards crisis protectionism? COVID-19 and selective deglobalization. Crit. Perspect. Int. Bus. 17 (2), 230–251.

Breitung, J., 2000. The local power of some unit root tests for panel data. In: Baltagi, B.H. (Ed.), Advances in Econometrics, Volume 15: Nonstationary Panels, Panel Cointegration, and Dynamic Panels. JAY Press, Amsterdam, pp. 161–178.

Charfeddine, L., 2017. The impact of energy consumption and economic development on ecological footprint and CO₂ emissions: evidence from a Markov switching equilibrium correction model. Energy Econ. 65 (1), 355–374.

- Chen, J., Xu, C., Cui, L., Huang, S., Song, M., 2019. Driving factors of CO₂ emissions and inequality characteristics in China: a combined decomposition approach. Energy Econ. 78 (1), 589–597.
- Chichilnisky, G., 1994. North-south trade and the global environment. Am. Econ. Rev. 84 (4), 851–874.

Choi, I., 2001. Unit root tests for panel data. J. Int. Money Financ. 20, 249–272.

- Chontanawat, J., 2020. Relationship between energy consumption, CO₂ emission and economic growth in ASEAN: cointegration and causality model. Energy Rep. 6 (1), 660–665.
- Civelli, A., Horowitz, A., Teixeira, A., 2018. Foreign aid and growth: a spatial PVAR analysis using satellite sub-national data for Uganda. J. Dev. Econ. 134 (c), 50–67. Coxhead. I., 2003. Development and the environment in Asia. Asian–Pacific Econ.
- Literat. 17 (1), 22–54.
- de Bruyn, S., van Den Bergh, J., Opschoor, J., 1998. Economic growth and emissions: reconsidering the empirical basis of environmental Kuznets curves. Ecol. Econ. 25, 161–175.
- Dinda, S., 2004. Environmental Kuznets curve hypothesis: a survey. Ecol. Econ. 49 (4), 431–455.
- Dinda, S., Coondoo, D., 2006. Income and emission: a panel-data based cointegration analysis. Ecol. Econ. 57, 167–181.
- Do Paço, A., Raposo, M., Leal Filho, W., 2009. Identifying the green consumer: a segmentation study. J. Target. Meas. Anal. Mark. 17 (1), 17–25.
- Drake-Brockman, J., 2018. Global Value Chain Policy Series: Environment. World Economic Forum.
- Duan, Y., Ji, T., Yu, T., 2021. Reassessing pollution haven effect in global value chains. J. Clean. Prod. 284, N.PAG.
- Dumitrescu, E.I., Hurlin, C., 2012. Testing for granger non-causality in heterogeneous panels. Econ. Model. 29, 1450–1460.
- Ee, M., Chao, C., Liu, X., Yu, E., 2018. Environmental policy, firm dynamics and wage inequality in developing countries. Int. Rev. Econ. Financ. 57, 70–85.
- Erdogan, A.M., 2014. Foreign direct investment and environmental regulations: a survey. J. Econ. Surv. 28 (5), 943–955.
- Fagerberg, J., Lundvall, B., Srholec, M., 2018. Global value chains, national innovation systems and economic development. Eur. J. Dev. Res. 30 (3), 533–556.
- Formai, S., Vergara Caffarelli, F., 2015. Quantifying the Productivity Effects of Global Value Chains, Working paper. University of Cambridge.
- Friedl, B., Getzner, M., 2003. Determinants of CO₂ emissions in a small open economy. Ecol. Econ. 45 (1), 133–148.
- Galeotti, M., Lanza, A., 1999. Richer and cleaner? A study on carbon dioxide emissions in developing countries. Energy Policy 27 (10), 565–573.
- Gereffi, G., 1999. International trade and industrial upgrading in the apparel commodity chain. J. Int. Econ. 48 (1), 37–70.
- Gibbons, S., Overman, H., 2012. Mostly pointless spatial econometrics. J. Reg. Sci. 52 (3), 172–191.
- Glucker, A., Driessen, P., Kolhoff, A., Runhaar, H., 2013. Public participation in environmental impact assessment: why, who and how? Environ. Impact Assess. Rev. 43, 104–111.
- Grossman, G., Krueger, A., 1991. Environmental impacts of a north American free trade agreement. In: NBER Working Paper 3914. NBER, Inc.
- Grossman, G., Krueger, A., 1995. Economic growth and the environment. Q. J. Econ. 110 (2), 353–377.
- Halkos, G., Tsionas, E., 2001. Environmental Kuznets curves: Bayesian evidence from switching regime models. Energy Econ. 23 (2), 191–210.

Hamilton, J., 1994. Time Series Analysis. Princeton University Press, NJ.

Hansen, L., 1982. Large sample properties of generalized method of moments estimators. Econometrica 50 (4), 1029–1054.

- Harris, R., Tzavalis, E., 1999. Inference for unit roots in dynamic panels where the time dimension is fixed. J. Econ. 91, 201–226.
- Hertwich, E., 2020. Carbon fueling complex global value chains tripled in the period 1995–2012. Energy Econ. 86 (1), 104651.
- Holtz-Eakin, D., Selden, T., 1995. Stoking the fires? CO₂ emissions and economic growth. J. Public Econ. 57 (1), 85–101.
- Holtz-Eakin, D., Newey, W., Rosen, H., 1988. Estimating vector autoregressions with panel data. Econometrica 56 (6), 1371–1395.
- Humphrey, J., Schmitz, H., 2002. How does insertion in global value chains affect upgrading in industrial clusters? Reg. Stud. 36 (9), 1017–1027.
- Ignatenko, A., Raei, F., Mircheva, B., 2019. Global value chains: what are the benefits and why do countries participate?. In: IMF Working Paper No WP/19/18.
- Im, K., Pesaran, M., Shin, Y., 2003. Testing for unit roots in heterogeneous panels. J. Econ. 115, 53–74.
- Ivarsson, I., Alvstam, C., 2010. Supplier upgrading in the home-furnishing value chain: an empirical study of Ikea's sourcing in China and South East Asia. World Dev. 38 (11), 1575–1587.
- Jaunky, V., 2011. The CO₂ emissions-income Nexus: evidence from rich countries. Energy Policy 39 (3), 1228–1240.
- Jawadi, F., Mallick, S., Sousa, R., 2016. Fiscal and monetary policies in the BRICS: a panel VAR approach. Econ. Model. 58 (C), 535–542.
- Jouanjean, M., Gourdon, J., Korinek, J., 2017. GVC participation and economic transformation: lessons from three sectors. In: OECD Trade Policy Papers No 207. OECD Publishing, Paris, France.
- Judson, R., Owen, A., 1999. Estimating dynamic panel data models: a guide for acroeconomists. Econ. Lett. 65 (1), 9–15.
- Kleemann, L., Abdulai, A., 2013. The impact of trade and economic growth on the environment: revisiting the cross-country evidence. J. Int. Dev. 25 (2), 180–205.
- Koopman, R., Wang, Z., Wei, S., 2014. Tracing value-added and double counting in gross exports. Am. Econ. Rev. 104 (2), 459–494.
- Kordalska, A., Parteka, A., Wolszczak-Derlacz, J., 2016. Global value chains and productivity gains: a cross-country analysis. In: Collegium of Economic Analysis Annals, Warsaw School of Economics, Collegium of Economic Analysis, 41, pp. 11–28.
- Kowalski, P., Gonzalez, J., Ragoussis, A., Ugarte, C., 2015. Participation of Developing Countries in Global Value Chains: Implications for Trade and Trade-Related Policies; OECD Trade Policy Papers No 179. OECD Publishing, Paris, France.
- Krueger, A., 1977. Growth, Distortions and Patterns of Trade among Many Countries. International Finance Section, Princeton, NJ.
- Kummritz, V., 2015. Global value chains: benefiting the domestic economy?. In: IHEID Working Papers 02–2015; Economics Section, the Graduate Institute of International Studies.
- Levin, A., Lin, C., Chu, C., 2002. Unit root tests in panel data: asymptotic and finitesample properties. J. Econ. 108, 1–24.
- Liu, H., Li, J., Long, H., Li, Z., Le, C., 2018. Promoting energy and environmental efficiency within a positive feedback loop: insights from global value chain. Energy Policy 121, 175–184.
- López, L., Arce, G., Zafrilla, J., 2013. Parcelling virtual carbon in the pollution haven hypothesis. Energy Econ. 39 (1), 177–186.
- Love, I., Zicchino, L., 2006. Financial development and dynamic investment behavior: evidence from panel VAR. Quart. Rev. Econ. Finan. 46 (2), 190–210.
- Lütkepohl, H., 2007. New Introduction to Multiple Time Series Analysis, 2nd ed. Springer-Verlag, Berlin.
- Lv, Y., Cui, X., Wang, D., 2019. GVC participation and carbon embodied in international trade: nonlinear analysis based on GMRIO and PSTR model. J. Quant. Tech. Econ. 36 (2), 45–65 (In Chinese).
- Magazzino, C., 2015. Economic growth, CO₂ emissions, and energy use in Israel. Int. J. Sustain. Dev. World Ecol. 22 (1), 89–97.
- Magazzino, C., 2016a. Economic growth, CO₂ emissions and energy use in the South Caucasus and Turkey: a PVAR analyses. Int. Energy J. 16 (4), 153–162.
- Magazzino, C., 2016b. CO_2 emissions, economic growth, and energy use in the middle east countries: a panel VAR approach. Energy Sources Part B Econ. Plan. Policy 11 (10), 960–968.
- Magazzino, C., 2016c. The relationship between CO_2 emissions, energy consumption and economic growth in Italy. Int. J. Sust. Energy 35 (9), 844–857.
- Magazzino, C., 2016d. The relationship between real GDP, CO₂ emissions and energy use in the GCC countries: a time-series approach. Cogent Econ. Finan. 4 (1), 1152729.
- Magazzino, C., 2017. The relationship among economic growth, CO₂ emissions, and energy use in the APEC countries: a panel VAR approach. Environ. Syst. Decis. 37 (3), 353–366.
- Magazzino, C., Cerulli, G., 2019. The determinants of CO₂ emissions in MENA countries: a responsiveness scores approach. Int. J. Sust. Dev. World 26 (6), 522–534.
- Meng, B., Peters, G., Wang, Z., Li, M., 2018. Tracing CO₂ emissions in global value chains. Energy Econ. 73 (1), 24–42.
- Meyer, K., 2017. International business in an era of anti-globalization. Multinatl. Bus. Rev. 25 (2), 78–90.
- Millimet, D., List, J., Stengos, T., 2003. The environmental Kuznets curve: real progress or misspecified models? Rev. Econ. Stat. 85 (4), 1038–1047.
- Moomaw, W., Unruh, G., 1997. Are environmental Kuznets curves misleading us? The case of $\rm CO_2$ emissions. Environ. Dev. 2, 451–463.
- Nickell, S., 1981. Biases in dynamic models with fixed effects. Econometrica 49 (6), 1417–1426.
- Oliver, J., Janseens Maenhout, G., Muntean, M., Peters, J., 2016. Trends in Global CO₂ Emissions; 2016 Report, the Hague: PBL Netherlands Environmental Assessment Agency. European Commission, Joint Research Centre, Ispra.

J. Wang et al.

Omri, A., 2013. CO₂ emissions, energy consumption and economic growth Nexus in MENA countries: evidence from simultaneous equations models. Energy Econ. 40 (1), 657–664.

- Parker, S., Bhatti, M., 2020. Dynamics and drivers of per capita CO₂ emissions in Asia. Energy Econ. 89 (1), 104798.
- Pesaran, M., 2004. General Diagnostic Tests for Cross Section Dependence in Panels. IZA Discussion Papers 1240. Institute of Labor Economics (IZA).

Pesaran, M., 2007. A simple panel unit root test in the presence of cross-section dependence. J. Appl. Econ. 22 (2), 265–312.

- Raupach, M., Marland, G., Ciais, P., Le Quéré, C., Canadell, J., Klepper, G., Field, C., 2007. Global and regional drivers of accelerating CO₂ emissions. Proc. Natl. Acad. Sci. U. S. A. 104 (24), 10288–10293.
- Reddy, K., Chundakkadan, R., Sasidharan, S., 2020. Firm innovation and global value chain participation. Small Bus. Econ. 1–21.
- Rodrik, D., 2018. New technologies, global value chains, and developing economies. In: NBER Working Papers 25164. NBER, Inc.
- Saggi, K., 2007. Chapter 10 Intellectual property rights and international technology transfer via trade and foreign direct investment. In: Maskus, K. (Ed.), Intellectual Property, Growth and Trade (Frontiers of Economics and Globalization, Vol. 2). Emerald Group Publishing Limited, Bingley, pp. 329–355.
- Sanyal, P., 2004. Intellectual property rights protection and location of R&D by multinational enterprises. J. Intellect. Cap. 5 (1), 59–76.
- Sbia, R., Shahbaz, M., Hamdi, H., 2014. A contribution of foreign direct investment, clean energy, trade openness, carbon emissions and economic growth to energy demand in UAE. Econ. Model. 36, 191–197.
- Selden, T.M., Song, D., 1994. Environmental quality and development: is there a Kuznets curve for air pollution emissions? J. Environ. Econ. Manag. 27 (1), 147–162.
- Shafik, N., 1994. Economic development and environmental quality: an econometric analysis. Oxf. Econ. Pap. 46 (Sup. 1), 757–773.

Sims, C., 1980. Macroeconomics and reality. Econometrica 48 (1), 1-48.

- Song, M., Wang, S., 2017. Participation in global value chain and green technology progress: evidence from big data of Chinese enterprises. Environ. Sci. Pollut. Res. 24, 1648–1661.
- Soytas, U., Sari, R., Ewing, B., 2007. Energy consumption, income, and carbon emissions in the United States. Ecol. Econ. 62 (3–4), 482–489.
- Stock, J., Watson, M., 2001. Vector autoregressions. J. Econ. Perspect. 15 (4), 101–115. Su, B., Ang, B., 2012. Structural decomposition analysis applied to energy and emissions:
- some methodological developments. Energy Econ. 34 (1), 177–188. Tobler, W., 1970. A computer movie simulating urban growth in the Detroit region. Econ. Geogr. 46 (1), 234–240.
- Toda, H., Yamamoto, T., 1995. Statistical inference in vector autoregressions with possibly integrated processes. J. Econ. 66 (1–2), 225–250.
- Tol, R., 2005. The marginal damage costs of carbon dioxide emissions: an assessment of the uncertainties. Energy Policy 33 (16), 2064–2074.

- Wang, C., Chen, J., Zou, J., 2005. Decomposition of energy-related CO₂ emission in China: 1957–2000. Energy 30 (1), 73–83.
- Wang, S., Fang, C., Guan, X., Pang, B., Ma, H., 2014. Urbanisation, energy consumption, and carbon dioxide emissions in China: a panel data analysis of China's provinces. Appl. Energy 136 (1), 738–749.
- Wang, J., Wan, G., Wang, C., 2019. Participation in GVCs and CO₂ emissions. Energy Econ. 84 (1), 104561.
- World Bank, 1992. World Development Report 1992: Development and the Environment. World Bank, Washington DC.
- WTO (World Trade Organization), 2018. Making Trade Work for the Environment, Prosperity and Resilience. WTO Press, Switzerland.
- Xu, S., He, Z., Long, R., 2014. Factors that influence carbon emissions due to energy consumption in China: decomposition analysis using LMDI. Appl. Energy 127 (1), 182–193.
- Yao, X., Shah, W., Yasmeen, R., Zhang, Y., Kamal, K., 2021. The impact of trade on energy efficiency in the global value chain: a simultaneous equation approach. Sci. Total Environ. 765, 142759.
- Yasmeen, R., Li, Y., Hafeez, M., 2019. Tracing the trade-pollution Nexus in global value chains: evidence from air pollution indicators. Environ. Sci. Pollut. Res. Int. 26 (5), 5221–5233.
- Ye, C.S., Ye, Q., Shi, X.P., Sun, Y.P., 2020. Technology gap, global value chain and carbon intensity: evidence from global manufacturing industries. Energy Policy 137, 111094.
- Yi, K., 2003. Can vertical specialization explain the growth of world trade? J. Polit. Econ. 111 (1), 52–102.
- You, W., Lv, Z., 2018. Spillover effects of economic globalization on CO₂ emissions: a spatial panel approach. Energy Econ. 73 (1), 248–257.
- Zhang, Y., Da, Y., 2015. The decomposition of energy-related carbon emission and its decoupling with economic growth in China. Renew. Sust. Energ. Rev. 41 (1), 1255–1266.
- Zhang, F., Gallagher, K., 2016. Innovation and technology transfer through global value chains: evidence from China's PV industry. Energy Policy 94, 191–203.
- Zhang, Y., Zhang, S., 2018. The impacts of GDP, trade structure, exchange rate and FDI inflows on China's carbon emissions. Energy Policy 120 (1), 347–353.
- Zhang, Z., Zhu, K., Hewings, G., 2017. A multi-regional input-output analysis of the pollution haven hypothesis from the perspective of global production fragmentation. Energy Econ. 64 (1), 13–23.
- Zheng, X., Yu, Y., Wang, J., Deng, H., 2014. Identifying the determinants and spatial Nexus of provincial carbon intensity in China: a dynamic spatial panel approach. Reg. Environ. Chang. 14 (4), 1651–1661.
- Zhou, R., Bi, X., Segerson, K., 2020. Evaluating voluntary environmental programs with spillover effects. J. Assoc. Environ. Resour. Econ. 7 (1), 145–180.
- Zhu, S., Tang, Y., Qiao, X., You, W., Peng, C., 2022. Spatial effects of participation in global value chains on CO₂ emissions: a global spillover perspective. Emerg. Mark. Financ. Trade 58 (3), 776–789.